

Received 28 August 2024; revised 14 September 2024; accepted 17 September 2024. Date of publication 20 September 2024; date of current version 22 October 2024.

Digital Object Identifier 10.1109/OJCOMS.2024.3465028

# A Review of UV Communications for UAVs XINGGUANG LI<sup>®</sup>, XINGLE XUE, CHEN ZHOU, JIANSHE MA<sup>®</sup>, AND PING SU<sup>®</sup> (Member, IEEE)

Division of Advanced Manufacturing, Shenzhen Graduate School, Tsinghua University, Shenzhen 518055, China CORRESPONDING AUTHOR: P. SU (e-mail: su.ping@sz.tsinghua.edu.cn)

This work was supported by the Shenzhen General Basic Program under Grant WDZC20220816110140002.

**ABSTRACT** As an important branch of free-space optical (FSO) communication technology, ultraviolet (UV) communication is mainly applied to mobile communication platforms represented by unmanned aerial vehicle (UAV). With the development of LEDs and UV detector devices, UAV UV communication technology has shown great potential in related fields. But at the same time, it also faces some challenges. As the communication distance increases, the path loss of the UV communication system can reach 0.12dB/m, and the variation in bit error rate (BER) can rapidly deteriorate from the order of  $10^{-8}$ to  $10^{-1}$ . Additionally, the UV communication system mounted on a UAV platform can emit radiation into the environment, which may have negative effects on human health when the radiation intensity exceeds  $0.5\mu$ W/cm<sup>2</sup>. These issues can be summarized as the availability, stability, and effectiveness of UAV-based UV communication technology. This paper aims to comprehensively address both UAVs and UV communication, providing a detailed introduction to the challenges and solutions facing UAV-based UV communication technology. Focusing on the specific aspects of these three issues, the paper first introduces the research background, value, and challenges of UAV-based UV communication technology, and investigates the current research status of UV communication channel models and positioning techniques. In order to solve the problem of UV environmental radiation, the article goes on to introduce beamforming and power control in UV optical communication technology. To solve the problem of reducing signal attenuation and increasing the communication range, the article introduces diversity technology and networking technology. In order to balance the communication quality and communication rate during UAV movement, the article introduces adaptive modulation and adaptive coding technology. Finally, the future development direction of UAV UV communication technology is summarised.

**INDEX TERMS** Beamforming, channel model, coding, diversity and combining, modulation and demodulation, networking, positioning techniques, power control, ultraviolet communications.

Ν	IOMENCLA	<b>FURE</b>	FOV	Field of View
	AOA	Angle of Arrival	FSO	Free-Space Optical Communication
	AWGN	Additive White Gaussian Noise	GNSS	Global Navigation Satellite System
	BER	Bit Error Rate	HD	Hardware Dimming
	BF	Bit-Flip	LDPC	Low-Density Parity-Check
	BP	Belief Propagation	LED	Light-Emitting Diode
	CRC	Cyclic Redundancy Check	LLRBP	Log-Likelihood-Ratio Confidence Propagation
	CSI	Channel State Information	LOS	Line-of-Sight
	DPIM	Differential Pulse Interval Modulation	MAC	Multiplexed Access Control
	DPPM	Differential Pulse Position Modulation	MIMO	Multiple Input Multiple Output
	DSTBC	Differential Space-Time Block Coding	MISO	Multiple Input Single Output
	EGC	Equal Gain Combining	ML	Maximum Likelihood Combining
	FAA	Federal Aviation Administration	ML	Maximum Likelihood

© 2024 The Authors. This work is licensed under a Creative Commons Attribution-NonCommercial-NoDerivatives 4.0 License.

For more information, see https://creativecommons.org/licenses/by-nc-nd/4.0/

MRC	Maximum Ratio Combining
MS	Minimum-Sum
NLOS	Non-Line-of-Sight
OOK	On-Off Keying
PDOA	Phase Difference of Arrival
PMT	Photomultiplier Tube
PPM	Pulse Position Modulation
RF	Radio Frequency
RS	Reed-Solomon
RSS	Received Signal Strength
SC	Selection Combining
SCL	Successive Cancellation List
SD	Software Dimming
SIMO	Single Input Multiple Output
SISO	Single Input Single Output
SMRC	Selection Maximum Ratio Combining
SNR	Signal-to-Noise Ratio
SOP	Secrecy Outage Probability
SSC	Switching-Dwell
STC	Space-Time Coding
TDOA	Time Difference of Arrival
UAV	Unmanned Aerial Vehicle
UV	Ultraviolet
VLC	Visible Light Communication
VMPPM	Variable Multi-Pulse Position Modulation
VMQAM	Variable Quadrature Amplitude Modulation
VOOK	Variable On/Off Keying Modulation
VPPM	Variable Pulse Position Modulation
ABER	Average Bit Error Rate
ACM	Adaptive Coding and Modulation
AI	Artificial Intelligence
AINM	Alternative Iterative Newton Method

# I. INTRODUCTION

AV HAS a wide range of applications in the military, agriculture, forestry, animal husbandry and service industries, and has gradually spawned the 'drone +' industry model, which has had a profound impact on the development of the world economy [1]. According to the Federal Aviation Administration's (FAA) estimates, the global production of drones will be about 2.7 billion, and by 2025, the production of civilian drones will increase to 10.9 billion. The drone industry will grow by an average of 15%-20% per year over the next decade and by more than 100% in 10 years [2]. The growth of the UAV industry relies on reliable product performance, and communication technology is a major limiting factor in UAV performance. Traditional UAV communications generally use radio frequency (RF) communications, but RF communications suffer from band congestion and are susceptible to eavesdropping and jamming, which limiting the application of UAVs in certain fields. Free space optical (FSO) communication has the advantage of not being subject to electromagnetic interference. It is a feasible solution for UAV communication, such as the UAV-satellite optical



FIGURE 1. Different Link Types for UAV UV Communication.

communication system [3] and the optical communication system using UAVs as auxiliary relays [4]. Meanwhile, the development of FSO communication technology is of great significance to the development of UAV low altitude economy [5].

In recent years, in the field of UAV space optical communication, UV communication technology has received more and more attention. UV communication uses UV light with a wavelength of 200nm to 280nm as a carrier wave, and mainly relies on the scattering of UV particles for signal transmission. With the development of UV LED light source and detector technology [6], [7], [8], [9], the practical value of UV communication has gradually emerged [10], [11], [12]. The basic UAV UV communication scenario is shown in Fig. 1.

Fig. 1 presents the basic scenarios of UAV UV communication according to the link structure, which is divided into UAV-UAV NLOS communication, UAV-UAV LOS networking, ground station-UAV LOS communication, and ground station-UAV NLOS communication. It is worth stating that UAV UV communication is not an alternative technology, but a supplement and improvement of the existing technology. Although UV communication is utilized in the airspace, traditional methods such as RF communication and fiber optic communication remain applicable for connections between ground stations. The reason for this choice is determined by the advantages and shortcomings of UV communication. Compared with traditional communication methods, UV communication has the following three advantages:

- Non-line-of-sight (NLOS) communications. UV particles are prone to interact with atmospheric particles and a NLOS communication link can be established through scattering. This property is suitable for communication over short distances and in complex terrain [13].
- Good confidentiality. The characteristic of UV light being easily scattered makes its signal-to-noise ratio (SNR) decay rapidly when the distance increases, and it is difficult to effectively eavesdrop information outside

the effective communication range, which is suitable for the safe networking of UAVs.

• Low background interference. UV communication uses UV light with a wavelength of 200nm to 280nm. In nature, the light in this wavelength is fully absorbed by the atmosphere, and there is little background interference, which makes it suitable for communication at any time of the day [14].

The above advantages of UV communication technology make it suitable for UAV-assisted communication, and it has a large scope in military, industrial and agricultural applications of UAVs. From a performance perspective, UV communication offers the following advantages over traditional methods such as RF communication and laser communication in terms of reliability, throughput, and performance under adverse conditions:

- Reliability: UV communication leverages its unique wavelength range (200 nm to 280 nm), which is absorbed by the atmosphere in the natural environment, effectively reducing environmental interference and the risk of signal eavesdropping [15]. This characteristic enhances the confidentiality and security of UV communication compared to RF communication. Furthermore, the use of diversity techniques and beamforming enables UV communication to maintain high signal quality in the presence of multipath effects and significant environmental noise, thereby improving communication reliability.
- Throughput: UV communication can achieve higher throughput through adaptive modulation and coding techniques that dynamically adjust the data transmission rate based on channel conditions. This flexibility allows UV communication to perform exceptionally well in scenarios with high data transmission demands. By combining multi-beam methods and appropriate MIMO coding, UV communication can achieve greater diversity gain, thus enhancing system throughput and demonstrating superior data transmission capabilities compared to traditional RF and laser communication methods.
- Performance under Adverse Conditions: In adverse environmental conditions (such as high noise and signal attenuation), UV communication can effectively control signal degradation by optimizing transmit power and adjusting beam radiation distribution, thereby maintaining good communication quality. This is particularly effective in complex terrains and short-range communication.

Despite the numerous advantages of UV communication for UAVs, it also faces several issues, including the tradeoff between data rate and error rate, the detrimental effects of UV environmental radiation, and limited communication range (as detailed in Section II). For example, the enhancement of UAV UV communication distance needs to take into account not only the communication quality but also the communication rate. But they can't be realised by unlimited increase of the transmitting power, so technologies such as beamforming, adaptive modulation and coding are also needed to realize the potentials of the existing systems. However, few comprehensive discussions on the above issues have been reported in the field of UAV UV communications. As shown in Table 1, [16], [17], [18], [19] gave an introduction to the characteristics and advantages of UV communication and related technologies, [14] described the application of UV communication technology in UAV networking, starting from the advantages of UV communication technology. These articles focus on UV communication technology and do not focus on the problems that mobile platforms represented by UAVs may encounter when combined with UV communication technology. Regarding the integration of space-based optical communication with aerial platforms, [20] has established a new hovering UAV-FSO channel model, taking into account atmospheric disturbances, turbulence effects, jitter, misalignment, reception errors, and wind resistance effects. Through the reception of incoherent signals at multiple positions, the spatial diversity-assisted non-FSO communication system significantly increased the system's redundancy and connection stability. Additionally, the study investigated UAV-FSO communication channel models under different performance metrics, considering the effects of engine vibration, tilt angle, and horizontal wind resistance on the UAVs. This provides a reference for the analysis of airborne UV communication from a performance analysis perspective.

In this paper, we will focus on the key technologies involved in applying UV communication technology to UAVs from the perspective of airborne UV communication technology. Focusing on the three issues mentioned above, we will provide an account of the current state of UAV UV communication, and introduce the problems and challenges in this field.

This paper will first describe the challenges faced by UAV UV communication technology. It will then introduce channel modeling and positioning techniques, which form the theoretical foundation for UAV UV communication. Then, power control and beamforming techniques in the field of UV communication will be introduced around the problem of background radiation in FSO communication scenarios. In order to solve the problem of limited communication distance, the introduction about diversity technology and networking technology will be carried out sequentially. On this basis, we hope to maximize the information rate under the premise of guaranteeing the communication quality in dynamic communication scenarios, so adaptive modulation and adaptive coding will be introduced. At the end of the article, we will present the future trends of UAV UV communication technology.

As shown in Fig. 2, the structure of this paper is as follows. Section II explains the challenges faced by UV communication technology. Section III introduces the channel model for UV communication including analytical

#### TABLE 1. Main content of existing research work.

Category	Main Content	References
	The latest advancements in NLOS UV communication systems include applications of artificial intelligence, artificial neural networks, game theory, and orbital angular momentum.	[19]
	The development history of UV communication, characteristics related to UV transceivers, and methods for improving UV communication system performance.	[16]
UV Communication	The research history and current status of UV communication, including the three main issues of channel models, system analysis and design, and optical sources and detectors, as well as future research directions in UV communication.	[17]
	The latest advancements in the technology and implementation of short-range UV communication links.	[18]
UV communication for UAV swarms	Network topology of UAV swarms based on UV communication, system Bit Error Rate (BER) performance, data rate, and communication range, as well as the challenges faced in UV drone network communication.	[14]



FIGURE 2. Structure of the paper.

model and Monte Carlo model, and Section IV shows the positioning techniques in UAV UV communication. Based on the two basic technologies, channel model and ranging and localization, more key technologies can be realized. Beamforming technology is introduced in Section V, power optimization technology is introduced in Section VI, diversity technology is introduced in Section VII, networking technology is introduced in Section VIII, networking technology is introduced in Section VIII, modulation technology is introduced in Section IX, and coding technology is introduced in Section IX, and coding technology is introduced in Section XI summarizes the main contribution of this paper with the development trend of UAV UV communication technology.

## **II. CHALLENGES**

UV communication technology has garnered significant attention for its advantages in UAV auxiliary communication. It offers substantial benefits in data transmission rates due to its higher spectrum resources and bandwidth. This technology can provide UAVs with stable and high-speed data transmission channels, thereby enhancing their operational efficiency and data processing capabilities in complex environments. However, the development of this field still faces some issues. As shown in Fig. 3, UV communication is used for transmitting information between UAVs and ground stations, as well as between UAVs. For the link between UAVs and ground base stations, the uplink transmits command information, while the downlink transmits status information and media information such as audio and video. Between UAVs. UV communication can transmit status information, media, and command information. During this information transmission, variations in environmental and system parameters can negatively impact communication distance, rate, and error rate. Overall, these negative impacts mainly include three aspects:

- Environmental radiation issues. Natural environments have limited resistance to UV radiation, so the UV energy radiated into the environment by the communication system needs to be controlled, which can be solved from three aspects. The first aspect is the selection of UV bands that are harmless to the human body [21], [22], [23], [24]. The second aspect is to optimise the transmit power and reduce it as much as possible under the premise of ensuring the communication rate and communication quality. The third aspect is to change the radiation distribution of the beam to concentrate the radiation region with the strongest SNR in the vicinity of the detector, which needs to be considered in the overall design of the performance indexes such as confidentiality, BER and communication rate.
- Limited communication distance. As the communication distance increases, the energy of the UV LED outgoing signal rapidly decays, and the SNR of the system rapidly decreases, making the effective communication distance of the single input single output (SISO) UV system limited [25], [26], [27]. In order to overcome this defect, three measures can be considered. The first is to increase the transmit power of the system



FIGURE 3. Key Technologies in UAV UV communication.

and try to establish line-of-sight (LOS) communication links with relatively low path loss and avoid NLOS links with high path loss. The second aspect is to use diversity and merging techniques to provide multiple independent fading channels in multiple dimensions, such as time, space and frequency, to increase the diversity of signals and thus reduce the impact of multipath effects and fading on the signals. The third aspect is the use of multi-hop communication or networking communication to improve signal coverage [28], [29].

 Trade-off between communication quality and communication rate during dynamic changes in UAV position. Due to the movement of the UAV, the communication link will vary between the shortest and the longest. But the path loss of the UV signal will increase rapidly with the increase of communication distance, which means the communication quality will also vary between the worst and the best. In order to optimise the communication rate while ensuring the communication quality, a coding and modulation method with higher information rate can be selected when the link distance is short. And a coding and modulation method with lower information rate but better BER performance can be selected when the link distance is long, which can be achieved by adaptive coding and modulation.

The division of the above issues doesn't have strict boundaries, and they are linked to each other. Adaptive coding and modulation are primarily used to balance BER and communication rate, while diversity and networking can enhance communication distance. Adaptive power control and beamforming techniques are used for controlling environmental interference. These technologies cover different areas but are all based on channel models and positioning technologies, as the latter provide essential status information for the practical application of other technologies. Next, this paper will begin with an introduction to the two fundamental technologies and proceed accordingly.



FIGURE 4. Development of channel models for UV communication.

### **III. CHANNEL MODELS**

The channel model is the basis of UV communication research. Based on the UV channel model, the system's BER performance, upper limit of communication rate, and parameters such as transmit power can be estimated. Therefore, the establishment of an accurate and effective channel model is the basis for the study of techniques such as channel estimation, channel coding and adaptive modulation and coding. There have been a large number of research on channel models under short-range, long-range, coplanar and non-coplanar conditions in UV communications, and some related research works are shown in Fig. 4. However, the

#### TABLE 2. Classification of channel models.

Classification	Content	References
	Coaxial single scattering model	[30]
	Non-coaxial single scattering model	[25]
Classification       Content         Coaxial single scattering model       Non-coaxial single scattering model         Analytical model       Circular truncated cone model         Empirical model of path loss       Non-coplanar single scattering model         Empirical model       Extended single scattering model         Improved non-coplanar single scattering model       Multiple scattering Monte Carlo model         Multiple scattering Monte Carlo model       Multiple scattering model in diversity reception         Multiple scattering Monte Carlo model       Multiple scattering model in diversity reception         Multiple scattering model carlo model in broad spectral range       Monte Carlo model carlo model in broad spectral range         Monte Carlo integration model       Effect of particle properties on Monte Carlo scattering model considering detector dead time         Monte Carlo scattering model considering typical obstacles       Monte Carlo scattering model considering height differences         Monte Carlo scattering model considering different receiver shapes       Spherical crown model for narrow beam case         Second-order and third-order scattering integral models       Single scattering turbulence model	[31]	
ClassificationContentCoaxial single scattering model Non-coaxial single scattering model Circular truncated cone model Empirical model of path loss Non-coplanar single scattering modelAnalytical modelExtended single scattering ellipsoid model Improved non-coplanar single scattering model Multiple scattering Monte Carlo model Improved multiple scattering model in diversity reception Multiple scattering model at multiple wavel Effect of particle properties on Monte Carlo scattering the Carlo scattering model considering typical obst Monte Carlo scattering model considering typical obst Monte Carlo scattering model considering different receive Spherical crown model for narrow beam case Second-order and third-order scattering integral model Single scattering turbulence model Monte Carlo omnidirectional scattering model	Empirical model of path loss	[32]
	ion         Content           aodel         Coaxial single scattering model Non-coaxial single scattering model Circular truncated cone model Empirical model of path loss Non-coplanar single scattering model           Extended single scattering ellipsoid model Improved non-coplanar single scattering model Multiple scattering Monte Carlo model Improved multiple scattering Monte Carlo model Multiple scattering Monte Carlo model Multiple scattering Monte Carlo model Multiple scattering Monte Carlo model Multiple scattering model in broad spectral range Monte Carlo multiple scattering model at multiple wavelengths Effect of particle properties on Monte Carlo scattering models Monte Carlo scattering model considering detector dead time Monte Carlo scattering model considering typical obstacles Monte Carlo scattering model considering height differences Monte Carlo scattering model considering height differences Monte Carlo scattering model considering height differences Spherical crown model for narrow beam case Second-order and third-order scattering integral models Single scattering turbulence model Monte Carlo omnidirectional scattering model	[33]
	Extended single scattering ellipsoid model	[34]
	Improved non-coplanar single scattering model	[35]
	Multiple scattering Monte Carlo model	[36]
	Improved multiple scattering Monte Carlo model	[37]
	Multiple scattering model in diversity reception	[38]
	Multiple scattering Monte Carlo model in broad spectral range	[39]
	Monte Carlo multiple scattering model at multiple wavelengths	[40]
Monte Carlo integration model	Effect of particle properties on Monte Carlo scattering models	[41]
	Monte Carlo scattering model considering detector dead time	[42]
	Monte Carlo scattering model considering typical obstacles	[43]
	Monte Carlo scattering model considering height differences	[44]
	Monte Carlo scattering model based on spherical coordinate system	[45]
	Monte Carlo scattering model considering different receiver shapes	[46]
	Spherical crown model for narrow beam case	[47]
	Second-order and third-order scattering integral models	[48]
	Single scattering turbulence model	[49]
	Monte Carlo omnidirectional scattering model	[50]

communication system in the context of UAVs still faces new challenges, such as the relative flight of UAVs, the effects of airframe vibration and altitude changes on the quality of communication, which are potential research directions for UV communication channel modeling.

According to the classification shown in Table 2, in Fig. 4, the green markers are the studies of analytical models of UV communication channels, and the blue markers are the studies of Monte Carlo scattering models. According to the existing studies, UV channel models can be classified into the following categories: coplanar models, non-coplanar models, approximate models and Monte Carlo models. In 1979, Reilly et al. established a single scattering model for the first time based on the ellipsoidal coordinate system, and analysed the time-domain characteristics of the singlescattering information transmission as well as the effects of Rayleigh scattering and Mie scattering. But the model was limited to the case of co-axiality of the transceiver and the transmitter/receiver [30]. To remove this limitation, in 1991, Luettgen et al. extended the single scattering model to the case of non-coaxiality at the receiver and transmitter and investigated the angular spectrum of the received energy, the impulse response and the path loss [25]. Both models [25], [30] contain multiple integrals, which are more complex in form and bring difficulties in studying the laws of the UV channel and in practical applications. In 2008, Xu et al. obtained a closed-form expression for calculating the path loss based on the single scattering theory of [31] by approximating the effective scattering volume as a circular table, and analysed the relationship between data transmission rate and distance. In order to obtain a simple and accurate channel model, in 2009, Chen et al. established an empirical model of path loss by fitting based

on outdoor experimental data and analysed the relationship between the communication performance and the transmit power, the communication distance, the pitch angle and the data transmission rate based on the model [32]. In terms of non-coplanar models: in order to extend the single scattering model to the non-coplanar case, in 2011, Elshimy et al. derived a single scattering model for the non-coplanar case based on the ellipsoidal coordinate system [33]. In the same year, Wang et al. proposed a closed-form expression for the single-scattering model considering the non-coplanar case based on the spherical coordinate system. The results showed that the results were compatible with those of [32] when the angle between the axes of the transceiver and the transmitter were small [34]. In 2013, Zuo et al. obtained a closed-form expression for the non-coplanar approximation model based on the non-coplanar exact model, which is more accurate compared to [33], [35].

The above UV channel model only considers the role of single scattering, with the increase of transmission distance, the role of multiple scattering becomes more obvious, and the law of multiple scattering is more complex, it is difficult to analyse it with analytical models. In 2009, Ding et al. used the idea of light tracing to establish a Monte Carlo simulation model for UV scattering communication, which takes into account the role of multiple scattering [36]. In 2011, Ding et al. improved the model of [36], and the improved model converges faster and is more robust, and it is able to efficiently capture the received energy contributed by different scattering times as well as the impulse response [37]. Due to the flexibility of the Monte Carlo model in application, the researchers also studied the diversity reception [38]; scattering communication performance at different wavelengths [39], [40]; the effect of aerosol density

and particle size on UV scattering communication based on the model [41]; the effect of photon counter dead time [42]; the effect of typical obstacles [43]; and the effect of height difference between receiver and transmitter [44].

In 2019, Wu and Shen et al. proposed a single scattering integral model based on the spherical coordinate system, including the structure of the effective scattering volume closed and open, and the geometrical rules for the division of communication scenarios [45], [46]. In order to reduce the computational complexity, Wu et al. established a single scattering approximation model for the case of small divergence angle [47]. In 2016, Yuan et al. developed the first integral model for secondary and tertiary scattering [48]. In 2020, based on the approximate model, Shan et al. investigated the effect of turbulence on the UV channel [49]. Since the analytical model is limited by a specific geometric structure, Yuan and Shen et al. improved the Monte Carlo integral model in 2019 and 2020, respectively, to improve the computational efficiency [51], [52]. In order to promote the practicality of UV communication, in 2020, Shen and Shan et al. investigated the receiver-transmitter jitter model for UV communication [53] and the omnidirectional communication model [50], which further improved the channel model for UAV UV communication. Reference [54] studied the path loss of non-line-of-sight (NLOS) UV communication systems in monodisperse and polydisperse aerosol systems, and found that the state of aerosol mixing has a significant impact on the performance of the communication system. Reference [55] has proposed an experimental system for precise measurement of UV channel and communication characteristics. The system comprises a transceiver system with a gimbal, UV light-emitting diode array, and photomultiplier tube detector, along with a LabVIEW-based data acquisition subsystem. Additionally, the article introduces novel techniques for accurately characterizing the radiation pattern of the UV LED array, absolute transmit power, and the field-of-view (FOV) of the detector.

For representing the model of UV communication channels, whether using analytical models or Monte Carlo models, the key to performance improvement lies in approximating irregular spatial geometries. For example, in the analysis of three scattering models in [56], their differences lie in the approximation methods of the scatterer. When the scatterer is approximated as a truncated cone, the path loss is given by:

$$L \approx \frac{96r \sin\theta_1 \sin^2\theta_2 \left(1 - \cos\frac{\phi_1}{2}\right) \exp\left[\frac{k_e r(\sin\theta_1 + \sin\theta_2)}{\sin\theta_s}\right]}{k_s P(u) A_r \phi_1^2 \phi_2 \sin\theta_s \left(12\sin^2\theta_2 + \phi_2^2 \sin^2\theta_1\right)}.$$
(1)

When the scatterer is approximated as a cuboid, the path loss is:

$$L \approx \frac{2\pi r \quad \sin\theta_1 \left(1 - \cos\frac{\phi_1}{2}\right) \exp\left[\frac{k_e r(\sin\theta_1 + \sin\theta_2)}{\sin\theta_s}\right]}{k_s P(u) A_r \phi_1^2 \phi_2 \sin\theta_s}.$$
 (2)



FIGURE 5. Three Approximations of Path Loss.

When the scatterer is approximated as cut and complement, the path loss is:

$$L \approx \frac{2\pi r^4 \sin^2\theta_1 \sin^2\theta_2 \left(1 - \cos\frac{\phi_1}{2}\right) \exp\left[\frac{k_e r(\sin\theta_1 + \sin\theta_2)}{\sin\theta_s}\right]}{k_s P(u) A_r V \sin^4\theta_s},(3)$$

*r* represents the distance between the transmitter and receiver,  $\theta_1$  and  $\theta_2$  are the elevation angles of the transmitter and receiver,  $\phi_1$  and  $\phi_2$  are the elevation field-of-view angles of the transmitter and receiver, and  $k_e = k_a + k_s$  is the scattering coefficient. Aris the area of the receiving detector, and P(u) is the scattering phase function, determined by the following formula:

$$\begin{cases} p(\cos\theta_s) = \frac{k_s^{Ray}}{k_s} p^{Ray}(\cos\theta_s) + \frac{k_s^{Mie}}{k_s} p^{Mie}(\cos\theta_s), \\ p^{Ray}(\cos\theta_s) = \frac{3[1+3\gamma+(1-\gamma)\cos^2\theta_s]}{16\pi(1+2\gamma)}, \\ p^{Mie}(\cos\theta_s) = \frac{1-g^2}{4\pi} \left[ \frac{1}{(1+g^2-2g\cos\theta_s)^{\frac{3}{2}}} + f\frac{3\cos^2\theta_s-1}{2(1+g^2)^{\frac{3}{2}}} \right]. \end{cases}$$
(4)

Simulations of path loss for the three aforementioned scenarios are shown in Fig. 5. From Fig. 5, it can be seen that the path loss when the scatterer is approximated as a cuboid is greater than when the scatterer is approximated as cut and complement, and the path loss when approximating the scatterer as cut and complement is greater than when approximating it as cuboid. This indicates that the closer the approximation method and mathematical representation of the common scatterer are to the actual situation, the higher the accuracy of the resulting model.

Research on UV static channel modelling is now relatively well developed, and the underlying logic of much of the work described in this chapter is based on the scattering theory of UV particles. Incorporating the UV communication channel model into a specific real-world scenario, considering more channel details, and aiming to improve the accuracy of the model are the general ideas of channel modelling research.

# **IV. POSITIONING**

In Section III, we introduced the development of the basic theory of UV communication channel model. This section will introduce the current research status of the basic technology, UV communication positioning technology. In techniques such as beamforming, power optimisation and adaptive modulation and coding, position information are the basis for algorithm execution. In UAV swarm network communication scenarios, accurate position and distance information is the basis for achieving efficient, safe and reliable formation operations. Positioning technology plays a key role in achieving efficient cooperative operations, avoiding collisions, and optimising resource utilisation [57]. The positioning of the communication system is based on a specific channel structure, regardless of visible light communication (VLC) or UV communication, the energy of NLOS channel is much lower than that of LOS channel, so most of the current mainstream positioning methods are based on LOS channel, and only some research in the field of UV communication will focus on considering the positioning technology under the condition of NLOS channel.

In the field of VLC, which is dominated by LOS channels, the accuracy of positioning techniques has reached centimetre-level [58], [59], [60]. LOS link-based positioning methods mainly achieve positioning through measurements of received signal strength (RSS), time difference of arrival (TDOA), phase difference of arrival (PDOA), and angle of arrival (AOA), which require multiple LOS links to estimate the user's location [61], [62]. The basic principle of the RSS method is that the received signal power decreases as the distance between the receiver and transmitter increases. According to this principle, it is possible to measure the distance in one-dimensional space and the position in two-dimensional space [63], [64]. However, the effects of obstacles and reflected signals mean that the relationship between distance and RSS is difficult to determine, which limits the accuracy of the RSS method. TDOA or PDOA utilise the time difference or phase difference of multiple received signals to make position measurements, and both they require the transmit signals to be perfectly synchronised in order to achieve a high level of accuracy in the measurement [64]. Currently, AOA is a more feasible method because it can achieve high positioning results through a relatively simple optical system [65].

However, the above studies are based on two-dimensional localization, and UAVs need three-dimensional spatial localization information. Generally speaking, three-dimensional localization to determine the height of the receiver. In [66], researchers divided the three-dimensional space into several two-dimensional planes, and determined the receiver's position in each two-dimensional plane based on the difference of the angular gain. The three-dimensional localization algorithm proposed in the paper achieved an average localization accuracy of 3.5 cm. In [67], Yang et al. proposed an indoor 3D localization system combining a single transmitter

with multiple tilted receivers, using the AOA method for 2D localization and RSS for 3D localization. The method achieved a positioning accuracy of within 6 cm in an indoor environment of 2m×2m×2.5m. In [68], Wang et al. proposed a 3D uplink indoor positioning system based on distributed receivers, which uses acceleration sensors to obtain angular information and does not require positioning assumptions about the mobile user's height or orientation angle. In [69], Gu et al. proposed assuming the height in the prediction stage and using nonlinear estimation in the correction stage to realize the 3D coordinate estimation of the height from 0.7 m to 1.7 m. In [70], Yin et al. used an angle diversity transmitter associated with an accelerometer for uplink 3D localization. In [71], Xu et al. used two ring receivers mounted with multiphotodiodes to localize the led light source. The design uses multiphotodiodes to calculate the angle, while the RSS method is used to calculate the distance, and the localization error is within 0.2 m. Regarding the positioning error, the current reports are mostly based on the GNSS-based UAV ranging and positioning accuracy level division: positioning accuracy within 5m is Healthy, positioning accuracy within 5m-13m is Slightly shifted, positioning accuracy within 13m-23m is inaccurate, and positioning accuracy more than 23m is dangerous [72].

The above is the localization technique in VLC based on LOS link, at present, there are fewer reports related to UAV UV communication localization technique on LOS link, and most of the studies are based on UV NLOS link. However, this does not mean that the performance of UV NLOS positioning technology is superior to LOS positioning technology, as the latter has smaller path loss at the same distance and can achieve higher accuracy. We can take an example of positioning in one-dimensional space to see how UV NLOS communication and LOS communication perform under different atmospheric extinction coefficients and different distances. The expression for the received optical power in the UV communication link is:

$$P_{r,LOS} = \frac{P_t A_r}{4\pi r^2} e^{-K_e r}.$$
(5)

By manipulating this formula, we obtain the distance calculation formula for LOS conditions:

$$r = \frac{2lambertw\left(\frac{1}{4}\sqrt{\frac{P_{t}A_{r}K_{e}^{2}}{\pi P_{r,LOS}}}\right)}{K_{e}}.$$
(6)

Similarly, the expression for the received power in NLOS communication is:

$$P_{r,LOS} = \frac{P_t A_r K_s P_s \phi_1^2 \phi_2 \sin(\theta_1 + \theta_2)}{32\pi^3 r \sin\theta_1 \left(1 - \cos\frac{\phi_1}{2}\right)} e^{\frac{K_e r(\sin\theta_1 + \sin\theta_2)}{\sin(\theta_1 + \theta_2)}}$$
(7)



FIGURE 6. The curves of positioning results varying with the extinction coefficient Ke. (a) r = 2m. (b) r = 10m. (c) r = 20m. (d) r = 50m.

By manipulating this equation, the distance from the transmitter to the receiver in NLOS conditions is:

$$r = \frac{lamberw\left(\frac{P_{t}A_{r}K_{s}K_{e}P_{s}\phi_{1}^{2}\phi_{2}(sin\theta_{1}+sin\theta_{2})}{P_{r,NLOS}32\pi^{3}sin\theta_{1}\left(1-cos\frac{\phi_{1}}{2}\right)}\right)}{\frac{K_{e}(sin\theta_{1}+sin\theta_{2})}{sin(\theta_{1}+\theta_{2})}}.$$
(8)

In the above four formulas,  $P_t$  represents the transmitted power,  $A_r$  is the aperture area of the receiver,  $P_s$  is the scattering phase function,  $K_s$  is the atmospheric scattering coefficient, and  $K_e$  is the atmospheric extinction coefficient.  $\phi_1$  is the FOV at the transmitting end,  $\phi_2$  is the FOV at the receiving end,  $\theta_1$  is the elevation angle at the transmitting end, and  $\theta_2$  is the elevation angle at the receiving end.  $P_{r,NLOS}$  is the received optical power. Based on Eq. (6) and Eq. (8), plot the UV LOS and NLOS positioning results at different communication distances as the extinction coefficient varies.

Set  $P_t = 200mW$ ,  $A_r = 0.001m2$ ,  $\phi_1 = 10^\circ$ ,  $\phi_2 = 20^\circ, \theta_1 = 30^\circ, \theta_2 = 30^\circ$ .  $K_e$  varies between 0.001 and 0.1, and the standard distances are 2m, 10m, 20m, and 50m, respectively, to obtain the positioning distance versus extinction coefficient curves shown in Fig. 6.

From Fig. 6, it can be observed that at shorter communication distances, the LOS positioning error and NLOS positioning error differ significantly, with their curves approximating parallel, indicating a clear advantage in precision for LOS positioning. As the communication distance increases to 10m, 20m, and 50m, both the LOS and NLOS positioning errors gradually increase and become more similar. This is because with the increase in communication distance, both the LOS and NLOS links experience significant signal attenuation, no longer meeting the centimeter-level positioning accuracy. At very long distances, even meter-level or even tens of meters of accuracy are difficult to achieve. However, since the transmitted power in the above simulation is set at 200mW, increasing the power would expand the range of accurate positioning. Based on the above analysis, it is evident that the advantage of UV positioning is limited to NLOS positioning within a certain range, and LOS positioning is not its primary characteristic. Therefore, researching UV NLOS positioning technology within a certain range is reasonable and necessary. Reference [73] has, for the first time, realized a pulsed UV laser positioning system based on 4H-SiC 4-QD. The system demonstrates high uniformity in performance across all four quadrants, including low dark current and high UV responsivity. By utilizing Charge Storage Amplifiers (CSAs) as an effective method for linear amplification of photocurrents, the system exhibits excellent linearity and high positioning accuracy. Furthermore, a prototype laser beam deflection angle self-calibration system has been developed, achieving low-error angle calibration. Reference [74] introduces a NLOS ranging method based on UV-C band communication, which does not require prior knowledge of the directions of the transmitter and receiver. Detailed sensitivity analysis is performed through Monte Carlo simulations, considering key parameters such as background noise, shot noise, photon sampling time, spacing between spatial samples, and steering angle errors. Based on this analysis, the article proposes a method for range estimation using UV-C links in complex GPS-denied NLOS scenarios. Reference [75] derived the geometric and physical constraints for NLOS UV positioning using linearly-arrayed receivers. Subsequently, the article derived the analytical relationship between the location parameters and pointing parameters of an unknown transmitter and proposed a NLOS UV positioning method with acceptable computational complexity. Simulation results indicate that the NLOS UV positioning method using photon-counting receivers can achieve a distance error less than 2 meters when the transmitting elevation angle is greater than 30 degrees and the separation distance is greater than 2 meters. Reference [76] proposed an UV NLOS positioning method based on a simplified single-scattering channel model, which can simultaneously obtain the positioning and pointing direction of the transmitter. Numerical results demonstrate that, under typical transceiver geometries within a 100-meter range, using two photon-counting receivers can achieve a positioning error less than 2 meters and an azimuth error less than 2 degrees. Additionally, the article found that the positioning performance can be improved by increasing the gap between the two receiving elevation angles. Reference [77] proposed a method of NLOS link incidence angle estimation based on the energy received by the detector array, similar to the RSS method in the LOS link localization technique, which requires that the transmitter is pointing at the receiver in a certain dimension. However, the transmitter in a real UV communication system can point in any direction, so there are major limitations in realizing localization using this method. Reference [78] designed a UAV swarm UV communication landing assistance system and studied the optical power distribution of UV LED arrays, the article based on the Lambert irradiance characteristics of UV LEDs. As the number of LEDs in the array at the transmitter end increases, the power distribution tends to be uniform. As the inclination angle of the receiver increases, the received optical power becomes smaller, and when the receiver angle is fixed, there exists a Lambertian order that maximizes the received optical power. The design is essentially a positioning technique similar to the MISO structure. Reference [79] then studied the UAV UV communication and positioning system with different structures, and investigated the SNR and the effect of the receiver and transmitter elevation angles of the SIMO system, the MISO system, and the MIMO system on the above mentioned indexes of the BER. The

results show that increasing the number of transceiver devices can effectively improve the quality of UAV swarm UV communication, which in turn improves the accuracy of ranging, localization and swarm control.

In addition to the methods that use LOS links or NLOS links alone to achieve localization, some researchers have also considered both LOS and NLOS scenarios to solve for the coordinates of the point to be localized. Based on the four-node localization method, [80] arranged the UV LEDs into a spherical structure and placed the omnidirectional receiver at the top of the spherical structure. The method can achieve more than 80% localization accuracy in a space with a length, width and height of 200m with an average error of less than 10m.

Distinguishing from the above localization algorithms, UV image localization is also an important localization method. Whether it is coarse localization by using the position change of UV light points in the image or precise localization by using the four-quadrant offset of the UV beam, these approaches have important applications on mobile platforms such as UAVs. Reference [81], [82], [83] proposed a fast localization method for indoor and outdoor UAVs with the help of UV LEDs and named the system UVDAR. The researcher proved through experiments that the system can guarantee the accuracy of relative positioning between UAVs, and the positioning error is less than 0.02 rad in a distance of more than 3m. Then, [84] proposed a wireless communication system based on UV camera based on the above UVDAR system, which decodes the original position information by processing the image captured by the UV camera. At the same time, the transmitter's position is tracked using Kalman filtering, and the transmitted signals from multiple UAVs are processed simultaneously using 4D Hough transform. UVDAR technology further enhances the effectiveness of UV positioning. Compared to traditional UV positioning technologies, UVDAR does not require complex communication networks and possesses strong environmental adaptability, maintaining stability and accuracy in various environments. Additionally, UVDAR is particularly suitable for formation control of multi-rotor UAVs, capable of guiding follower UAVs to maintain given three-dimensional positions and orientations relative to the lead UAV. Successful validations under stringent outdoor conditions have paved the way for the application of UVDAR in actual multi-UAV formations. The innovative significance of UVDAR technology in UV communication positioning lies mainly in its provision of a new method for relative positioning between UAVs without the need for complex communication infrastructure. By utilizing UV light for realtime control, it offers new possibilities for UAV formations to operate within small teams. Overcoming the limitations of traditional positioning technologies in complex environments, UVDAR technology provides greater flexibility and reliability for UAV applications in various settings through the use of the UV spectrum for relative position and yaw measurements.



FIGURE 7. Battlefield Search and Rescue.

To more specifically demonstrate the advantages of UV communication, we have chosen the case of battlefield search and rescue, where there are two types of positioning needs: outdoor wide-area positioning and indoor smallarea positioning. The positioning methods for drones are set as UV LOS positioning, UV NLOS positioning, VLC positioning, WIFI communication, and 5G radio frequency communication positioning, respectively. The optimal communication method needs to be selected under different conditions. As shown in Fig. 7, UAV1-UAV4 are conducting search and rescue operations indoors, while UAV5-UAV7 are conducting search and rescue operations outdoors.

• If the enemy is not conducting RF signal interference at this time, then UAV1-UAV7 can all choose 5G RF communication for positioning, with a positioning accuracy of 2m-3m. However, for indoor positioning, if higher accuracy is required, it is necessary to further select WIFI positioning technology. But when the walls of the building are thick, it is difficult to establish a stable connection with WIFI positioning (as shown in Fig. 7, the signal generated by the WIFI router is difficult to continue propagating to UAV3 after passing through the walls). In this case, the NLOS positioning technology of UV communication can be used as a supplementary means. At the same time, due to factors such as the ease of WIFI signal leakage and difficulty in implementation, the feasibility of using WIFI positioning is not high. Therefore, UV positioning technology becomes the best choice for indoor NLOS situations (as shown in Fig. 7, the mutual positioning between UAV2 and UAV3).

If the enemy is conducting RF signal interference at this time, then UV communication, VLC, or infrared communication can be chosen for outdoor operations. If it is necessary to avoid the light source being monitored (for example, visible light should be avoided at night), UV positioning technology can be selected. The positioning accuracy can reach the centimeter level, but it is limited to small-scale communication and positioning (i.e., the positioning technology used by UAV4 in Fig. 7). When a larger communication range is required, infrared positioning technology with lower path loss can be chosen, achieving centimeter-level accuracy (i.e., the positioning technology used by UAV5 in Fig. 7). However, the drawback of infrared positioning is that it cannot achieve NLOS positioning. In large-scale NLOS positioning scenarios, UV positioning technology is still needed. If avoiding the monitoring of the light source is not necessary, in large-scale LOS communication scenarios, visible light positioning technology with even lower path loss can be chosen.

The aforementioned scenarios demonstrate the advantages of UV communication technology in indoor and outdoor positioning, which can be summarized as the capability to achieve NLOS positioning and high confidentiality. To make the comparison clearer, we summarize the characteristics of UV communication and conventional communication, as shown in Table 3.

As shown in Table 3, various positioning methods have their own characteristics. RSS positioning is simple but limited by channel conditions; TDOA and PDOA have high accuracy but are complex to implement; AOA relies on known channel information and has low expandability. Visual positioning equipment is simple but has limited accuracy; UV communication positioning includes LOS, NLOS, and combined positioning, with strong adaptability but requires LOS conditions; UV image positioning is efficient but cannot adapt to NLOS situations. 5G RF and WiFi positioning technologies are mature but have limited accuracy; LIDAR positioning is accurate but costly. Generally, each method has its unique advantages and limitations, and the choice of the appropriate positioning technology should be based on the specific application scenario, environmental conditions, and accuracy requirements. However, another approach is to promote the integrated development of various technologies. For the topic of this article, two types of integrations can be considered: RF communication positioning + UV communication positioning, and WiFi communication positioning + UV communication positioning. Table 4 shows the effects that can be achieved with these two integrated technologies. VOLUME 5, 2024

This section presents the methods and characteristics of UV communication positioning, and compares its advantages over traditional positioning methods such as VLC and RF communication. This is achieved through the use of NLOS signals that do not rely on reflective surfaces for positioning. This advantage is both a feature of UV communication and its positioning technology. However, this does not mean that UV communication positioning technology is superior to traditional positioning technology in all cases, as both have their own application scenarios and advantages. The former is suitable for NLOS scenarios such as indoor search and rescue, small-scale communication on the battlefield. etc., while the latter is more suitable for conventional communication scenarios. Therefore, in future research, the NLOS advantages of UV communication should be fully utilized to optimize existing LOS and NLOS positioning technologies. At the same time, UV communication, as a supplement to traditional communication methods, combines with traditional positioning technologies (such as GPS, WiFi, RF, etc.), which is another important research direction. This is because the fusion of multi-source information can significantly improve the accuracy and reliability of positioning. In addition to the above two aspects, with the development and maturation of UV detection devices, the miniaturization, noise resistance, resolution, and sensitivity of large-scale UV detector arrays are gradually improving, which provides the possibility for the practical application of UV image positioning technology.

#### **V. BEAMFORMING**

Although UV communication has the advantage of high confidentiality due to its high path loss, in UAV UV communication scenarios, there is still the risk of eavesdropping due to the random movement of the transceiver. In MIMO and network communication scenarios, interference between links is a more difficult problem than the eavesdropping risk. Therefore, we hope that the transmitter can generate directional beams with specific parameters so that the SNR near the receiver is high and in the area outside the receiver is low, thus minimizing the risk of eavesdropping. At the same time, we also want to minimize inter-link interference in MIMO and network communication scenarios. In order to achieve the above goals, beamforming technology is one of the solutions that can be used [93].

In the field of adaptive antenna, the beamforming technique weights and synthesizes the vector signals generated by multiple antenna array elements to suppress the non-target direction signals and form the target direction signals. And in the field of intensity modulation and direct detection of FSO communication, its channel is a Gaussian scalar channel, not a vector channel, so beamforming is generally realized by controlling the optical power [94]. Unlike conventional beamforming, the tuned signal driving the LED must satisfy certain current amplitude constraints to avoid distortion during the switching process. So current amplitude constraints rather than power constraints are usually imposed when assigning shape to the LED beam [95], [96]. The design of the

## TABLE 3. Characteristics of different positioning methods.

Name	Principle	Accuracy	Advantages	Disadvantages	Application Scenarios	References
VLC RSS	Distance is estimated by measuring the power of the received wireless signal, which is inversely propor- tional to the distance. The receiver collects RSSI values and converts them into distance estimates after comparing with reference values.	Error is less than 5cm within a range of 60cm×60cm.	Simple and easy to implement in hardware.	High requirements for channel conditions; ac- curacy is low when there are floating particles or obstacles in the channel.	Indoor short-range position- ing, such as smart homes and indoor navigation.	[85]
VLC TDOA	Distance is estimated by the time difference of arrival of signals emitted from a light source at the receiving end.	Error is less than 5.3cm within a space of 50cm×50cm.	Does not require the use of atomic clocks for high-precision synchronization, reducing reliance on high-	Difficult to implement.	Indoor positioning requir- ing higher precision, such as high-precision position- ing systems in unmanned factories.	[86]
VLC PDOA	The direction of signal arrival is determined by using the phase dif- ference of the same signal received by multiple receivers, and the sig- nal's direction of origin is calcu- lated based on the relationship be- tween phase difference and angle.	Average 2D tracking error is 4.3 cm, and average tracking er- ror in 3D tracking scenarios is 9.3 cm.	devices. High precision.	Depends on a high-precision synchronization system and analog- to-digital conversion devices, high cost, and high computational complexity of the	Situations requiring high- precision positioning, such as precision instrument con- trol.	[87]
VLC AOA	Uses trigonometry to locate by de- termining the angles between the user device and multiple position- ing anchors.	Angular resolution is less than 1 degree when the distance between the detector and the light source is orester than 3cm	High precision.	Relies on known chan- nel information, low ex- pandability.	Situations requiring deter- mination of the direction of signal origin, such as indoor positioning and navigation.	[88]
VLC Visual positioning	Uses a camera or image sensor to receive VLC signals, thereby ob- taining the ID and position infor- mation of the LED.	The minimum error can reach 6.5cm.	Simple equipment, low cost.	Limited precision, and each target object's posi- tion is encoded to create a graphic code.	Situations requiring high- precision positioning, such as precision instrument con- trol and robot navigation.	[68]
					(Co	ntinued)

# TABLE 3. Characteristics of different positioning methods (Continued).

Name	Principle	Accuracy	Advantages	Disadvantages	Application Scenarios	References
UV NLOS Link Positioning	Utilizes the characteristics of UV NLOS channels to calculate the distances between multiple light sources and detectors, and then determines the detector's position based on spatial geometric rela-	Error less than 3 degrees in static scenes, and less than 7 degrees in dynamic scenes.	Can achieve non-line-of-sight positioning that does not rely on reflective surfaces.	Requires the transmit- ter to point in one di- mension towards the re- ceiver, limiting the use of scenarios.	Outdoor positioning in complex environments such as cities and forests.	[77]
UV LOS and NLOS Link Combined Positioning	Estimates the position of the re- ceiver by combining LOS and NLOS channel information. Processes the images captured by	Error below 10m within a range of 200m.	Wide application range, strong anti-interference capability.	Depends on accurate channel models.	Suitable for indoor and out- door positioning requiring high precision and strong adaptability.	[80]
UV Image Positioning	the UV camera to decode the orig- inal position information. Simulta- neously, uses Kalman filtering to track the pose of the transmitter and uses 4D Hough transform to process the transmission signals from multiple drones simultane-	Achieves stable com- munication connec- tions in a 320-degree horizontal field of view and a 110- degree vertical field of view.	Can process multiple positional information in parallel, high efficiency.	Unable to adapt to non- line-of-sight situations.	High-precision positioning scenarios such as drone nav- igation, robot positioning, etc.	[84]
5G RF-based Localization	Estimates the position of the re- ceiver using RSS, TOA, TDOA, and AOA, etc.	Typically between a few meters to tens of meters.	Mature technology, relatively low cost, easy to deploy.	Affected by multipath effects, signal attenua- tion, and occlusion, with limited precision.	Suitable for high-speed transmission and high- precision positioning scenarios such as autonomous driving,	[06]
WiFi Com- munication Positioning	Determines the position by mea- suring the received wireless signal strength (RSSI) or signal propaga- tion time.	Several meters in in- door environments, and has poor accu- racy in outdoor envi- ronments.	Simple deployment, low cost, wide signal coverage. High precision.	Affected by signal interference, with unstable positioning accuracy, and high power consumption.	Suitable for indoor short- Suitable for indoor short- range positioning, such as smart homes, indoor navi- gation, etc.	[91]
Laser LIDAR Positioning	Measures the distance between the target and the sensor through laser scanning, combined with camera or radar images to determine the position.	Up to centimeter level, especially in short distances.	not affected by radio interference, suitable for complex environments.	High cost, sensitive to environmental light	autonomous driving, robot navigation, etc.	[92]

TABLE	4.	UV communication i	integrated	positioning	method.
	-		megrateu	positioning	methou.

Positioning Methods	Characteristics
RF Communica- tion Positioning	<ul> <li>Advantages: Mature technology, widely used, with a broad coverage area. Relatively low cost, easy to deploy.</li> <li>Disadvantages: Limited positioning accuracy due to multi- path effects and signal attenuation. Highly susceptible to environmental inter- ference and obstructions.</li> </ul>
UV Communica- tion Positioning	<ul> <li>Advantages:</li> <li>High precision positioning over short distances.</li> <li>Strong anti-interference capability, less affected by radio signal interference.</li> <li>Disadvantages:</li> <li>Relatively low technology maturity, higher cost.</li> <li>Limited coverage area, mainly used for specific regions.</li> </ul>
WIFI Communication Positioning	<ul> <li>Advantages:</li> <li>Wide signal coverage, simple deployment.</li> <li>Supports high-speed data transmission, suitable for applications with large data volumes.</li> <li>Disadvantages:</li> <li>Positioning accuracy is greatly affected by the environment, easily interfered with.</li> <li>High power consumption, not suitable for continuous operation over long periods.</li> </ul>
RF Communi- cation and UV Communication Fusion Positioning	Combines the broad coverage of RF com- munication with the high precision position- ing advantage of UV communication. Improves the robustness and adaptability of the positioning system, especially in com- plex and changing environments. Increases in cost and technical challenges, requiring more complex data fusion algo- rithms.
WIFI Communication Positioning and UV Communication Fusion Positioning	Combines the wide coverage and high-speed data transmission capabilities of WIFI com- munication, along with the high precision positioning advantage of UV communica- tion. Suitable for indoor and outdoor positioning, especially in scenarios requiring high-speed data communication. Higher system complexity, greater cost and technical challenges.

beamforming system is mainly based on this principle in VLC, which is dominated by the LOS channel, and in UV communication, which is dominated by the NLOS channel.

In the field of UV beamforming, the researchers proposed an indoor UV dynamic beamforming system based on positioning technology, which utilizes the user's position information in the current phase to realize beamforming in the next phase. However, the study was conducted based on the UV LOS channel, and beamforming in the UV NLOS channel has not been considered [97]. In the same year, Sudhanshu Arya et al. considered the LOS link and NLOS link in MIMO channel, and modeled the downlink beamforming problem as a second-order cone planning problem to maximize the SNR at the receiving end [98]. In addition to this, little research has been reported on UV beamforming, in contrast, beamforming research in the field of VLC is more abundant. Based on the similarity between UV communication and VLC in terms of hardware model, LOS channel and information processing, we can get some inspirations from beamforming technology in VLC field [99].

The first is the use of beamforming to enhance the confidentiality of the signal transmission process. Wyner has shown in 1975 that when the channel capacity between the target receiver and the transmitter is larger than the capacity of the eavesdropping channel, the confidential transmission of the signal can be realized [100]. But the exact expression of the secrecy capacity is complicated. In 2015, AymanMostafa et al. studied the secrecy and LED beamforming problem of SISO system and MISO system. The article pointed out that it is difficult to get the analytical expression of the secrecy capacity even in the simplest SISO case. Therefore, in a certain fixed-form communication scenario, it is generally assumed that the eavesdropper exists in a specific region and that the transmitter does not have access to the full channel information of the eavesdropper. The problem is ultimately modeled as maximizing a lower bound on the confidentiality performance of the communication link in each legitimate channel [94]. In the same year, the team investigated the beamforming problem for LED arrays, where the amplitude and polarity of the signal are used to shape the overall radiation pattern of the LED array, pointing the main lobe towards the intended receiver while minimizing the signal level in other regions. The problem was modeled as a linear optimization problem with moderate complexity, and the worst-case secrecy rate [101] was given. Next, the team investigated the design of beamformers in MISO eavesdropping channels, and the problem was reduced to maximizing the secrecy rate under an amplitude constraint. However, the original problem is nonconvex, so the article transformed it into a quasi-convex optimization problem and solved the beamforming weight vector that maximizes the secrecy rate [102]. Compared with the power constraint, the amplitude constraint describes the LED more accurately, which makes the above work also an important reference value for the study of the confidentiality of UV communication.

In addition to the modulation of the carrier, the distribution characteristics of the signal itself have an effect on the performance of beamforming. The related team investigated the secrecy rate of the input signal under amplitude constraints for different input distributions [103], [104]. Zaid et al. improved on the above research work by analyzing the secrecy of the system at truncated Gaussian input distribution and found that the effect is better than the secrecy when it consists of uniform inputs. The article also estimated the eavesdropper's channel using location information and used maximizing the secrecy rate lower bound as a strategy for beamforming [103]. Arfaoui et al. investigated the achievable secrecy rate of various input signal distributions including truncated generalized normal, Gaussian, and uniform distributions in a beamforming scheme considering artificial noise. The truncated generalized normal distribution has a significant effect on the secrecy rate improvement compared to the other two distributions [104].

In addition, the beamformer, which is centered on network precoding, is also an important factor in the design process. Pham et al. investigated the design of forced-zero precoding for a multiuser MISO broadcast channel. In the research, convex optimization modeling of the optimal beamformer is carried out based on upper and lower bounds on the capacity of the amplitude-constrained Gaussian channel. Using the precoding design, the bounds on the maximumminimum fairness and maximum sum rate are derived. The results show that the generalized inverse design has better performance than the pseudo-inverse design [105]. Pham et al. continued to study the precoding design in the case of multiuser active and passive eavesdropping, and designed the artificial noise-assisted precoding and forced-zero precoding schemes to improve the confidentiality of the system [106]. Arfaoui et al. proposed a new linear precoding scheme and its iterative algorithm respectively, which outperforms the traditional forced-zero precoding scheme in terms of confidentiality performance. However, the time complexity of the algorithm proposed in the article has increased, which is a detail that can be improved in future beamforming precoding work [107].

In UAV communication scenarios, the position between the target channel and the eavesdropping channel often varies, and the positional distribution of eavesdroppers is also a factor affecting the secrecy of the beamforming system. Reference [108] analyzed the secrecy outage probability (SOP) when eavesdroppers are randomly distributed using stochastic geometric methods. The paper improved the security performance of the system by minimizing the average SNR of the eavesdropper (ED) and maximizing the SNR of the target user. Reference [109] applied a Poisson distribution model to model the random location of eavesdroppers. The article considered several configurations of transmitters and receivers with different numbers and distributions, and derived a closed form for the probability of secrecy disruption in each case. In the research, the secrecy performance of the downlink is improved based on the LED selection scheme of the legitimate user's location. In addition, in [110], the researchers analyzed the secrecy performance in the presence of random colluding eavesdroppers.

Above is a brief introduction about the use of beamforming technology to enhance the system confidential. The beamforming design in the field of UAV UV communication is still in its infancy. The channel of UV communication is dominated by NLOS, and the FOV, pointing angle and link distance of the transceiver will affect the distribution characteristics of the LEDs and the channel gain [111]. In the design of the UV beamforming system, we need to consider the influence of the above system parameters, instead of using the Lamb distribution characteristics as the main basis for the channel gain in the above research work. In addition, the analytical representation of the UV NLOS channel model is difficult to obtain, and the existing work is based on the Monte Carlo integral model, which generates a high time complexity. Therefore, in the design of UV beamforming with convex optimization modeling as the main means, one of the key issues is to obtain a concise expression of the channel gain, and then the beamforming problem is expressed or approximated as a convex optimization problem, so as to obtain a computationally efficient beamforming algorithm [112]. At present, practical UV communication band is concentrated in the UVC band, which has a narrower linewidth and a higher requirement for the use of frequency division multiplexing to reduce interlink interference. In the above study, a precoding scheme is considered to improve the confidentiality and inter-link interference, which is undoubtedly an important reference value for the link interference and confidentiality of UV MIMO systems and network communications.

#### **VI. POWER OPTIMIZATION**

On the one hand, the path loss of UV signals is serious, and the energy at the receiver decreases significantly with the increase of communication distance, so it is necessary to enhance the transmit power of the signal to ensure the quality of communication. On the other hand, the external environment has limited resistance to UV radiation, and it is necessary to control the UV power below a certain threshold value. Therefore, the power control strategy that reasonably weighs the transmit power and the communication performance index is an important link in UV communication [16], [18].

Currently, research work on LED emitted power control is concentrated in the field of VLC, which can be categorized into hardware dimming (HD) [113], [114], [115], [116], [117], [118], software dimming (SD) [119], [120], [121], [122], [123] and hybrid dimming [124], [125], [126], [127], [128].

HD relies on the actual physical variables in the communication system for power control, and realizes the regulation of optical and electrical power by controlling variables such as LED on-current, number of conductors, LED optical axis pointing, and divergence angle [124]. HD subsystems are often nearly independent of data transmission subsystems and are a simple and efficient solution [115]. According to the different application scenarios, HD can be divided into HD for SISO systems, HD for MIMO systems, and HD for broadcasting systems, and most of them are modeled as convex optimization problems that can be solved quickly [116]. In the modeling process, in addition to the aforementioned transmitter parameters, the receiver pointing angle can also be used as a control variable. Although the difficulty of hardware implementation has increased, the power control accuracy has been improved [113]. There are also optimization designs that aim to maximize the received energy, such as the optimization problem of an integrated communication and energy supply system, in which a suitable objective function needs to be designed to balance communication quality and energy transmission [117]. As for the HD of broadcasting systems, the power allocation scheme is a major concern, and related studies have demonstrated that the optimal symbol modulation power allocation scheme possesses lower energy consumption than the equal power allocation scheme [114].

SD controls the duty cycle and modulation depth of the modulating signal to achieve the optimization of optical or electrical power with no change in hardware parameters. For specific implementation, Variable Pulse Position Modulation (VPPM), Variable On/Off Keying Modulation (VOOK), Variable Multi-Pulse Position Modulation (VMPPM) and Variable Quadrature Amplitude Modulation (VMQAM) can be used as the specific solutions for SD [120], [123]. For low-order modulation methods such as VPPM and VOOK, when controlling the transmit power by adjusting the signal duty cycle, the carrier frequency needs to be higher than a certain value, otherwise the BER performance of the dimming system is not significantly improved compared to the original system. In contrast, the use of modulation depth to control the transmit power offers superior flexibility [122]. For higher-order modulation methods such as VMPPM and MQAM, the adjustment of the signal duty cycle is the main part of power control, but the demodulation of higher-order modulated signals requires a high degree of smoothness in the signal energy. So in addition to balancing the signal quality and the transmit power, the trade-off between the resolution of the power control and the accuracy of the signal demodulation is a much more important issue [119], [121]. The advantages of SD are real-time and low cost, and the disadvantages are limited dimming effect, which needs to be combined with HD to achieve the desired effect in airborne dynamic communication scenarios.

Hybrid dimming integrates the characteristics of HD and SD to adjust the electrical parameters of the LED array, the optical structure and the parameters of the modulation signal under the limitations of optical power, band utilization, BER and other indicators. In addition to the adjustment of the structure of the transmitter, the change of the angle of the receiver and the appropriate demodulation algorithm can further mitigate the negative impacts of environmental noise, multipath dispersion and frequency distortion, so as to obtain higher spectral efficiency, lower detection error and more uniform power change curve [126], [128]. Hybrid dimming can further expand the power dynamic range of HD and SD, and enhance the robustness of the system in complex scenarios. It is an excellent dimming method.

The previous research relies on the known LOS channel gain, but the channel characteristics of UV communication, in which NLOS is the main application scenario, are difficult to be expressed in closed form. So its optimization method cannot be directly applied to UV communication. Therefore, it is valuable to study power control schemes applicable to UV communication systems [129]. In the current work, [112] proposed an alternative iterative Newton method (AINM) to optimize the repeater position and transmission power for full-duplex relay-assisted multihop UV communications with respect to UV NLOS characteristics. However, the study failed to point out the convexity of the established model, so the optimization algorithm used is the computationally inefficient alternating descent method. Therefore, establishing the power control problems in different kinds of UV communication systems into clearer models, using the information of the models themselves such as derivatives to provide the execution basis for the optimization algorithms. and trying to improve the accuracy of the optimization models and the execution efficiency of the optimization algorithms is a direction for future development in this field.

#### **VII. DIVERSITY**

Atmospheric scattering effect is the most important transmission mechanism of UV communication system, however, there is not only path loss, but also slow fading and fast fading in the signal transmission process. Among them, fast fading comes from the strong scattering of UV light in the atmosphere, and slow fading is generated by the undulation of light intensity caused by atmospheric turbulence [130]. Within a distance of a few tens of meters, the effect of atmospheric attenuation on the system performance can be ignored, and the system performance is mainly affected by the environmental noise. However, when the channel distance is hundreds of meters or more or when there are more suspended objects, the joint effect of the two needs to be considered. The use of diversity techniques can effectively improve this situation [131].

Diversity technology, which is the use of multiple receivers in a communication system, can mitigate the effects of atmospheric attenuation and environmental noise on the system, and is widely used in FSO [132], [133]. At present, the mainstream diversity techniques mainly include space diversity, frequency diversity and time diversity. In UV communication system, the cost of time diversity is too high, and the realization of frequency diversity is difficult, so the most widely used is spatial diversity technology [134]. In practical spatial diversity systems, the spatial correlation between UV beams can significantly degrade the performance of the system, so effective aperture separation is the key to achieving maximum diversity gain [135]. However, the relative concentration of different links will increase the inter-link interference, which is contradictory to the enhancement of diversity performance. So it is necessary to design suitable algorithms to take both into account, and realize the effective merging of signals from different links [136]. The physical structure design scheme of space diversity technology is limited, and the merging algorithm of signals has become an important direction in the study of space diversity technology in UV communication.

Signal merging techniques appropriately merge multiple statistically independent fading signals received, thereby reducing the effects of fading and improving system performance [137]. Currently commonly used merging algorithms include maximum ratio combining (MRC), equal gain combining (EGC), selective combining (SC), and maximum likelihood combining (ML) [138]. Reference [139] analyzed EGC diversity reception technique and validates it with experimental results. Unlike EGC, MRC compensates for channel differences between each photomultiplier tube (PMT) receiver. Based on the original MRC algorithm, [140] proposed a MRC algorithm that can compensate the channel difference between each PMT for UV communication systems. Relative to the EGC algorithm, this method can produce a larger output power, which effectively reduces the BER of the system. Subsequently, [141] investigated the effects of transmission distance and transmitter FOV angle on the EGC merging technique in UV single-inputmultiple-output (SIMO) systems, and the results showed that the EGC algorithm can produce better diversity gain under small transmitter FOV and short link conditions. However, when the FOV angle at the transmitter is too large or the distance is too far, the EGC provides little improvement in BER. Reference [142] proposed a selective maximum ratio combining algorithm (SMRC), and conducted experiments at distances ranging from 5 to 35 meters. The results show that the SMRC algorithm reduces the BER by 32% and 46% compared to the MRC and EGC algorithms, respectively, at a link distance of 30 m.

Diversity system can improve the performance of the system, but it does not mean that the more number of receivers is better. For the number of receivers, [143] investigated the BER of UV NLOS communication systems under different number of receivers and elevation angles of the transceiver. The results show that the BER performance of the four-receiver system is significantly better than that of the two-receiver system under the same distance and receivertransmitter elevation angle conditions. Reference [144] demonstrated that in UV systems, when the number of receivers reaches a certain value, the effect on the system performance is negligible. The link structure, on the other hand, has a large impact on the diversity performance, and we would like the receivers to be far enough apart from each other to obtain a better reception performance. However, in UV communication systems, the distance between receivers is often limited and the system may also operate in NLOS conditions. For the UV NLOS link structure, [145] derived the ML diversity detection scheme for spatially correlated UV NLOS links based on [132]. Simulation results show that ML merging outperforms SC and EGC when the SNR are the same in weak turbulence environments, and SC and EGC outperform ML in strong turbulence environments. However, SC and EGC merging methods are able to include some or all of the state information of the channel, which is not available in ML merging [146]. Similarly, two reception techniques, SC and ML, have been studied

for UV communication systems. Reference [147] studied the BER performance of SC and ML in weak turbulence environment. The results show that ML and SC schemes can significantly improve the BER performance of the system, and the performance of ML is due to SC in weak turbulence environments. Reference [148] proposed an M-PSK based NLOS switching-dwell receiver diversity technique for the UV NLOS link, and analyzed the relationship between the BER and the modulation depth. The article points out that the average bit error rate (ABER) of the system decreases as the modulation parameter M increases. Reference [149] investigated the secrecy of NLOS systems, and by comparing the secrecy of a single-link system with that of a system to which the EGC and MRC diversity techniques are applied. It was found that the diversity technique improves the secrecy performance of the system with a small off-axis error.

Regarding diversity in UV MIMO systems, [150] investigated different diversity reception schemes for diversity in  $2 \times 2$  MIMO systems under dispersive and non-dispersive channel conditions, respectively. The results show that the use of multibeam approach can significantly improve the throughput of the system. In addition to that, proper MIMO coding can achieve higher diversity gain. Space-time coding (STC) is a typical representative of MIMO coding technique, which uses a two-dimensional time interval and spatial coding method to encode the original sequence into a redundant sequence, which is then transmitted in parallel using multiple transmit antennas [151]. Reference [152] improved one of the simplest Alamouti codes in STC and applied it to the MIMO system of NLOS, and the results showed that the scheme can effectively reduce the system BER. However, the Alamouti code used in this study requires real-time channel state information (CSI) of the UV channel, so its practical application is greatly restricted [153]. In order to compensate for this shortcoming, [154] studied the application of differential space-time block codes (DSTBC) in  $2 \times 2$  MIMO UV communication systems, and the results showed that DSTBC coding does not need to obtain realtime CSI in time-varying fading UV channels, which is a great advantage of DSTBC over Alamouti codes. Next, [155] applied the symmetric optical pulse position-orthogonal space-time block code (SOPP-OSTBC) to a  $2 \times 2$  MIMO system with a ML merging scheme at the receiver. The results showed that the design outperforms its combination with the 2PPM modulation method when combined with the OOK modulation method. Reference [156] compared the performance of three merging methods, SC, EGC and MRC, under BPSK modulation, and the results showed that the use of MRC merging enables the system to obtain the maximum communication distance.

The above studies show that for a fixed diversity merging method, there should be an optimal modulation method, and similarly, for a deterministic modulation or coding method, there is an optimal diversity merging method. Of course, the trade-off between different methods needs to be based on a deterministic system structure. The current research on diversity techniques in MIMO systems focuses on two major categories, SIMO and  $2 \times 2$  MIMO. So extending the existing research to MIMO systems above the second-order is a key research direction in the future.

Overall, in UV communication systems, diversity techniques widely adopted include spatial, frequency, and time diversity, with spatial diversity being most commonly utilized due to its relative simplicity and cost-effectiveness. An analysis of existing technologies and research allows us to forecast potential trends and directions for future research in diversity technologies for UV communication systems:

- Enhanced Design of Diversity Receivers: Research into more effective diversity reception technologies is needed to improve the stability of signal reception and reduce BER. This includes optimizing the layout of receivers and designing more efficient reception algorithms to better cope with atmospheric scattering and turbulence.
- Integration of Multi-beam and MIMO Technologies: Combining multi-beam with MIMO technologies can significantly enhance the throughput and signal stability of UV communication systems. Future research should explore innovative applications in this area, especially under NLOS communication conditions.
- Optimization of Coding and Modulation Techniques: Current research has demonstrated the impact of various coding and modulation techniques on diversity performance. Future studies might further investigate how to optimize these techniques to suit different environments and communication distances, particularly selecting the most appropriate coding and modulation technologies for UV communication systems.
- Accurate Acquisition of CSI: Certain diversity techniques, such as the Alamouti code, require real-time CSI, which can limit their application under some conditions. Future research could explore methods and algorithms for efficient signal merging and coding without relying on real-time CSI.
- Optimization of Link Structure and Receiver Configuration: Research indicates that the relative positioning of link structure and receivers significantly impacts diversity performance. Future studies might explore how to optimize these parameters to improve the overall system performance and reliability.

Exploration and research in these directions can enable diversity technologies in UV communication systems to better adapt to complex and variable communication environments, thereby enhancing system stability and efficiency and leading to breakthroughs in future communication technologies.

#### **VIII. NETWORKING**

The high path loss of the UV channel makes the communication distance of the UV SISO system limited, and the networking technology can effectively improve this situation. As shown in Fig. 8, in complex environments with obstacles,

6514

UV networking communication can effectively expand the coverage of the original communication link. In addition, in the field of FSO communications such as VLC and laser communications, FSO networking technology is seen as a measure to solve the last kilometer of the communication network [157]. Regarding the general concepts and principles of UVC networking, [158] proposed requirements such as NLOS, not easily eavesdropped, and not requiring strict direction alignment, as well as the concept of an unattended ground transmission network. Reference [159] analyzed enemy and friendly aircraft links in three different scenarios, and proposed the concept of maximum standoff distance (i.e., the maximum distance that can simultaneously ensure that a friendly aircraft can be detected and an enemy aircraft cannot be detected). The authors pointed out that the channel gain, detection probability and maximum standoff distance all decrease as the communication distance increases. Apart from that, most of the related studies have been conducted on access control protocols and the structure of UV networks.

In UV communication networking, Multiplexed Access Control (MAC) plays a key role in defining how the network shares data during transmission, and efficient MAC protocols are essential for improving network performance [160], [161]. As shown in Fig. 9, there are three main types of current MAC protocols [162]. The first type is the competitive MAC protocol based on random access, which has lower requirements for the coordination mechanism and can adapt well to changes in the network configuration. In competitive MAC protocols, data streams share a common channel. When there is a conflict between UAV A and UAV B, the protocol enters a competitive phase, and the winner of the competition has the right to send information first. However, as the amount of node information increases, conflicts during transmission will increase, making the network performance significantly degraded [163], [164]. The second category is the non-competitive MAC protocol, which allocates a dedicated channel to each node and eliminates the competition for channel resources. In non-competitive MAC protocols, data sent by different UAV nodes have dedicated channels, which can be specific frequencies, specific times, or specific code words. Compared to the random access mechanism, non-random MAC protocols provide better channel utilization and reliability in high network traffic states [165]. The third category is the hybrid MAC protocol, which combines the advantages of both competing and non-competitive MAC protocols. Hybrid MAC protocols can improve the utilization of channel resources when the traffic load is low like competitive protocols, avoid conflicts when the network traffic increases like non-competitive type protocols, and solve the problem of synchronous allocation of network resources [166]. Reference [167] proposed a oneway ant colony routing algorithm based on the traditional two-way ant colony routing algorithm, which selects the path with the optimal performance in terms of delay and bandwidth, and the overall connectivity and survivability of the network is greatly improved. Reference [168] took



FIGURE 8. Expansion of Communication Range through Multi-Point Networking Technology for UVC in UAV.

obstacle avoidance between nodes and improvement of coverage and connectivity as the optimization objectives, and proposed a path selection algorithm for UV communication networking in three-dimensional space. Compared with the random-type MAC protocol, the coverage of the network and the average convergence time of the algorithm are both improved. Reference [169] designed a multipoint topology access protocol for star-shaped UV networks, which has a large improvement in the normalized throughput compared to the CSMA/Ca-like protocol, and is suitable for the combination of telecommunication networks and UV communication sub-networks. Reference [170] proposed an adaptive node assignment and power allocation method subject to capacity and node connectivity constraints for hierarchical UV scattering networks. This method can improve the communication rate of the worst node by more than 10%.

Compared with the research on MAC protocols for UV communication networks, the research on network structure configuration is more extensive. Reference [171] proposed a hierarchical area coverage optimization algorithm based on genetic algorithm, which improves the coverage of the network compared with the existing coverage algorithms, but the method is only applicable to the deployment in coplanar area. Reference [172], on the other hand, proposed a mesh non-coplanar MIMO structure based on linear geometry, which has lower reuse cost. The article also improved the original spherical decoding algorithm, and the improved algorithm reduced the complexity while ensuring the same computational accuracy. Reference [173] proposed a diamond-shaped structure based on the single-scattering model of UV NLOS links. Compared with the squarestructured network, the diamond-shaped network sacrificed the effective coverage area to improve the connectivity, and the performance index of the network was yet to be further improved. These studies are dominated by

node configuration, but the node configuration method is only one aspect of the network structure configuration. Reference [174] also studied the coverage radius and communication performance of NLOS UV communication networks in arbitrary polygonal regions. The results showed that, compared to the node model in quadrilateral regions, the node model of the network uniformly distributed in a circular region requires more stringent communication parameters. And the smaller the elevation angle of the transceiver is, the better the connectivity performance of the network is. On the other hand, [175] investigated the effects of changes in node density, transmit power, and data rate on the connectivity of UV communication networks for UAVs. The article pointed out that the connectivity of the network increases with the increase of node density and transmit power, and decreases with the increase of data transfer rate. The number of network nodes required for uniform distribution is lowest for the same connectivity.

Connectivity and coverage area are important parameters describing the performance of the network structure, and inter-link interference is an important criterion for evaluating the stability of its performance. The analysis and suppression of inter-link interference is of great significance for UV communication networking [176]. Reference [177] deduced the nature of the distribution of interference and its upper and lower bounds based on the Poisson clustering process of network nodes. The article also considered the probability of successful transmission of a signal in an interferencelimited channel in the context of the Rayleigh fading model. Reference [178] proposed a topology control algorithm for the channel interference problem in three-dimensional kconnected UV communication networks. The algorithm first generates an initial k-connected network, then streamlines the network structure, removes redundant parts, and finally uses a particle swarm optimization (PSO) algorithm to obtain an approximate optimal power allocation scheme. The improved



Hybrid MAC address protocol

#### FIGURE 9. Three types of MAC protocols.

algorithm not only generates a fault-tolerant topology, but also significantly reduces the inter-channel interference, which is suitable for networks with high stability requirements. Reference [179] proposed a new topology control algorithm, the main purpose of the algorithm is to construct a k-connected topology graph, which allows the network to remain connected when k-1 nodes fail. The network structure generated by this algorithm is able to control the node interference value to a stable value that does not increase with the number of nodes. Reference [178] proposed a low-interference fault-tolerant networking algorithm based on the 3D k-YG algorithm and the PSO algorithm [180], which can effectively reduce the interference in the UV communication network and improve the stability of the topological network. Reference [181] improved the link weight function for the residual energy and channel loss of communication nodes, and proposed a three-dimensional spatial communication network generation algorithm based on the optimal rigid graph. The network structure generated by this method can reduce the energy loss of the system and has good connectivity.

The related team has also explored the transceiver parameters and the number of nodes in the network that make up the UVC network. Reference [182] investigated the effects of the number of nodes, transmit power and data rate on the performance of the k-node network under different transceiver elevation angles. The results show that the smaller the transceiver elevation angle is, the larger the node communication range is. Reference [183] designed a geometric model for UAV ground-to-air UV communication networking. In this geometry, multiple transmitters are evenly distributed on the ground and the UAV receives signals in the air. The number of transmitters required by the system is inversely proportional to the horizontal distance between the receiving and transmitter, and the maximum coverage of the network can be obtained while meeting the BER metrics. Reference [184] investigated the BER and minimum irradiance after interference compensation on the basis of [183]. The article pointed out that the BER gradually increases with the number of network interference sources and the receiver field of view angle.

Besides, UV UAV self-organizing network technology is also an important direction. Mobile self-organization technology is based on some unique UV network characteristics, such as constantly changing topology and nodes within the network being both routing and destination nodes [185]. Most of these characteristics are constraints on the network nodes, and for a particular node in the network, a neighbor discovery algorithm is the basis for achieving its interconnection with surrounding nodes [186]. In the field of traditional communication, there have been related studies on neighbor discovery algorithms with known information about surrounding nodes [187], [188] and neighbor discovery algorithms with unknown a priori information [189], [190], [191]. However, none of these studies can be directly used in UV communication due to the differences in channel characteristics.

In recent years, there has been extensive attention on the access mechanisms for UV self-organizing networks. Research in this area primarily falls into two categories: random access protocols based on Time Division Multiple Access (TDMA) and random access protocols based on contention.

For TDMA protocols, time is divided into periodic frames, each frame further subdivided into several time slots, with fixed slots assigned according to certain allocation principles. However, since nodes in a self-organizing network are constantly changing, dynamic TDMA should be used, where time slots are dynamically competed based on node priorities. In 2016, researchers proposed a new dynamic TDMA protocol that improves performance by reallocating node priorities to reduce collisions among nodes with the same priority [192]. In 2023, a new TDMA-based random access cooperative network protocol was proposed, combining the advantages of synchronous and asynchronous media access control protocols. This protocol achieved higher throughput with lower latency and packet loss rates through a dynamic slot allocation mechanism [193]. Additionally, a 2023 study proposed a dynamic distributed TDMA protocol based on theoretical analysis and network simulation verification, using a fully connected matrix of nodes for distributed scheduling [194]. In 2024, to accommodate large-scale UV networks, researchers proposed a UV MAC protocol combining clustering algorithms with TDMA, featuring clustering mechanisms and dynamic slot allocation characteristics [164].

In networks with a large number of nodes, clustering algorithms are undoubtedly an effective method to enhance performance. In 2016, Yu et al. proposed a novel mobile prediction clustering algorithm for dynamic selforganizing networks to address the adaptability issues of traditional clustering algorithms in high-dynamic large-scale networks [195]. In 2017, Zeng and Zhang introduced a Low Energy Adaptive Clustering Hierarchy (LEACH) algorithm [196]. That same year, Wu et al. proposed an optimized clustering algorithm based on LEACH, which effectively reduced network energy consumption and extended network lifespan by considering the remaining energy of nodes in wireless sensor networks [197]. Reference [162] proposed a UVOC-MAC protocol specifically designed for outdoor UV communication networks. The article first introduced the impact of signal propagation and the physical layer on medium access through experiments, and established a noncoplanar path loss model to describe signal attenuation. Then, it described in detail the working process of the UVOC-MAC protocol, including idle state, sending and receiving of RTS and CTS messages, data transmission, etc. Specifically, the core steps of the UVOC-MAC protocol include:

- Initialization: The node initializes and maintains relevant tables, such as connection tables and occupancy tables.
- Data transmission: When the node has data to send, it first checks the connection table and occupancy table to select the best direction and pointing angle.
- Request to Send: The node sends an RTS message containing pointing angle information, thereby implicitly conveying the selected data rate.
- Clear to Send: After receiving the RTS, the receiving node checks whether it can receive data and sends a CTS message, which contains the highest possible data rate.
- Data transmission: The sending node sends data based on the data rate in the CTS. If possible, it uses fullduplex mode.
- Update tables: The node updates relevant tables after sending or receiving data.

These steps ensure that the UVOC-MAC protocol can efficiently utilize spatial dimensions, reduce collisions, and adapt to changing network environments. Subsequently, the article analyzed the effectiveness of the UVOC-MAC protocol in providing high throughput and reducing collisions through simulations. In addition, the article compared the UVOC-MAC with traditional MAC protocols used in radio frequency communications, emphasizing its uniqueness and applicability in outdoor UV communication environments.

In self-organizing networks, the dynamic nature of node positions leads to continuous changes in network topology, making neighbor discovery algorithms crucial as they are a prerequisite for random access protocols. Reference [198] considered the stability characteristics of UV communication network nodes, and utilized multi-feature fusion and GRF-MAP methods to sense the distributed dynamic network topology, and realized a dynamic adaptive routing protocol. In 2011, Wang et al. proposed a UV neighbor discovery algorithm based on the leader's dominant capability. During neighbor discovery, only the leader can send signals, continually optimizing to ensure all nodes in the network can equally become leaders [199]. In 2015, Yang et al. introduced a new TDMA-based UV communication directional neighbor discovery algorithm. This algorithm uses a backoff delay mechanism based on directional transmission and reception to quickly adapt to topology changes and resolve node collision issues [200]. In 2016, Zhang et al. developed a multi-channel UV communication directional channel access protocol based on spatial division multiplexing, establishing a UV communication network model for aircraft and addressing the "deafness" problem in UV directional communication in flight formations [201]. Reference [202], on the other hand, used an evaluation mechanism to rank the possible transmission directions of nodes, and then proposed a neighbor discovery protocol for UV self-organizing networks that can improve the search performance based on the information of node transmission directions. Assuming that the distribution of self-organizing network nodes obeys a Poisson process, [170] investigated the connectivity probability of the network and the minimum power for the network to remain connected under different antenna parameters. In addition, the article considered a two-tier network using directional antenna arrays and designed node assignment and power control algorithms. Reference [186] proposed a neighbor discovery algorithm based on probabilistic separation, which uses scheduling reference techniques to recommend nodes and avoids direct references to nodes in clusters, achieving a balance between neighbor discovery efficiency and energy efficiency.

Considering the dynamic nature of nodes in selforganizing networks, node mobility can impact network performance, necessitating the study of network connectivity during node movements. In 2018, Zhao et al. explored UV communication technology suitable for UAV networks, discussing network connectivity under random waypoint (RWP) and circular motion models [175]. In 2019, a study on UAV UV communication under circular uniform distribution conducted simulation experiments on the effects of elevation angles, node density, transmission power, and data rates on network connectivity (k=1,2) [182]. A 2020 study analyzed the impact of elevation angles of receiving and transmitting ends on network connectivity in UV communication network models when nodes move according to the RWP motion model, finding that irregular polygonal area models fit reality better than prototype models, reflecting the dynamic movement characteristics of network nodes [203].

In other research details, [204] improved the formation coordination algorithm for UV communication UAV swarms. The results show that the stability of the UAV swarm tends to stabilize with the increase of flight time. Reference [205] proposed an obstacle avoidance algorithm for UV communication of UAV swarm on the basis of the augmented vector field histogram algorithm, which can effectively prevent collision between UAV swarms. Reference [206] proposed a double ascent algorithm to determine the optimal position of the relay node for UV communication in the case of obstacle obstruction, which is superior to the traditional Bellman-Ford algorithm.

In the context of UV communication networking, it is essential to focus on both the reliability and rate of individual links, as well as the overall performance of the network. Directional antenna arrays enhance signal strength and quality through concentrated transmission and reception, effectively mitigating signal attenuation caused by scattering. By employing beamforming techniques, directional antenna arrays can optimize the propagation direction of signals, thereby reducing interference and multipath effects, which ultimately improves transmission efficiency. These characteristics enable them to provide more stable and efficient signal transmission in UV communication.

As previously discussed, constructing directional antenna arrays and designing node allocation algorithms for UV communication networks necessitates the selection of highgain antennas suitable for the UV spectrum, as well as the adoption of linear, planar, or three-dimensional array configurations to optimize signal transmission direction and minimize interference. Additionally, beamforming techniques should be utilized to adjust the amplitude of UV signal transmission antennas to enhance the quality of signals directed towards target users, while adaptive algorithms must be designed to respond to environmental changes. Moreover, the rational design of network topology and node density is crucial for network performance. The optimization of the aforementioned parameters helps to enhance the network's connectivity and throughput, ensuring the efficient operation of the ultraviolet communication network.

On the other hand, node allocation algorithms aim to optimize the layout and power distribution of nodes based on the characteristics of UV communication networks. A welldesigned layout and allocation mechanism can effectively reduce interference between signals, thereby enhancing transmission efficiency and increasing the overall data throughput of the network. The design of node allocation algorithms should be grounded in network modeling and utilize optimization algorithms, including swarm intelligence algorithms such as genetic algorithms and particle swarm optimization, as well as convex optimization techniques, to determine the optimal positions and power allocations of nodes, ensuring network coverage and connectivity while maintaining dynamic adaptability. Proper node allocation can significantly enhance network connectivity and throughput, ensuring efficient and reliable UV communication in complex environments.

In summary, in UV communication networking, the network must dynamically adapt to the continuously changing topology and node roles through various methods. Adaptive routing algorithms (such as AODV, OLSR, etc.) adjust data transmission paths in real-time to accommodate the addition or removal of nodes. Topology discovery and maintenance use topology control protocols to regularly update network topology information, ensuring that all nodes are aware of the latest network structure. Load balancing mechanisms optimize resource usage and improve network performance by dynamically adjusting task and data flow distribution, especially during changes in node roles. Node role assignment implements flexible role allocation mechanisms (such as master and slave nodes, coordinators and worker nodes), dynamically assigning tasks and responsibilities based on node capabilities and network needs. Additionally, fault tolerance and recovery mechanisms establish robust fault-tolerant strategies and rapid recovery plans to address node failures or anomalies, ensuring network stability and reliability.

### **IX. MODULATION**

The modulation method of a communication system should be adapted to the characteristics of the signal itself (wavelength, intensity, etc.) and the transmission environment (path loss, occlusions). Although there are often multiple modulation methods for a signal, in practice the choice of the best modulation method for the current transmission environment is limited [207].

As shown in Fig. 10, the SNR of UV communication, VLC, and infrared communication are simulated in a conventional atmospheric environment. The theoretical basis for this simulation is the photon scattering theory based on Mie scattering and Rayleigh scattering. Mie scattering refers to the scattering phenomenon that occurs when light waves encounter particles comparable in size to their wavelength, applicable to larger particles, with scattering intensity related to the size, shape, and refractive index of the particles. Rayleigh scattering, on the other hand, describes the scattering phenomenon when light waves interact with particles much smaller than their wavelength, primarily affecting short-wavelength light (such as blue light). The relationship between scattering intensity and wavelength for Mie scattering is more complex, while the intensity of Rayleigh scattering is inversely proportional to the fourth power of the wavelength, and both phenomena jointly affect the propagation of light.

The wavelength of UV light is shorter than that of visible light, and the wavelength of visible light is shorter than that of infrared light. Except for the fact that the wavelength of UV light is comparable to the diameter of particles in the air, the carrier wavelengths of the other two traditional communication methods are much larger than the diameters



FIGURE 10. Comparison of the three communication methods.

of air particles. In other words, the Mie scattering effect for UV particles is more pronounced, while the Mie scattering effects for the other two types of particles are relatively weak. Therefore, under the same modulation scheme, we set the Mie scattering coefficient for VLC to be one-tenth of that for UV communication, and the Mie scattering coefficient for infrared communication to be one-hundredth of that for UV communication. We then performed simulations on the system's SNR, resulting in the outcomes shown in Fig. 10 [56]. In the simulation, the transmission power is set to  $Pt = 30 \times 10^{-3}$ ; the atmospheric model parameters are  $k_{sr} =$  $0.24 \times 10^{-3}$ ,  $k_{sm} = 0.25 \times 10^{-3}$ , and  $k_a = 0.9 \times 10^{-3}$ . The area of the receiving window is  $A_r = 1.77 \times 10^{-4}$ . The model parameters are  $\gamma = 0.017$ , g = 0.72, and f = 0.5. The transmission end elevation angle  $\theta_T = 60$ , the receiving end elevation angle  $\theta_R = 60$ , the transmission end FOV angle  $\phi_T = 60$ , and the receiving end FOV angle  $\phi_R = 60$ . The results indicate that in short-range scattering communication scenarios, specifically NLOS communication scenarios, UV communication has a higher SNR compared to traditional communication methods, leading to a lower BER and a higher achievable transmission rate.

The wavelength of most UV communication systems is concentrated in the C-band between 200nm and 280nm. Compared to RF communication, infrared optical communication and VLC communication, UV has shorter wavelengths and stronger energies, which require higher modulation devices [13]. Traditional modulation mechanisms include electro-absorption modulation, electro-optical modulation and photoacoustic modulation. Electro-absorption modulation utilizes the absorption and emission of electrons and holes in the semiconductor to achieve modulation of the optical signal. Electro-optical modulation uses an electric field to control the refractive index of the material to achieve modulation of the optical signal [208]. Optoacoustic modulation utilizes the photoacoustic effect, in which the propagation of a light wave in a medium causes acoustic waves in the medium, to achieve modulation of the optical signal [209]. Unlike common radio frequency communication and fiber optic communication, the wavelength of UV light is shorter, and the particle nature is more pronounced (commonly described by Mie scattering and Rayleigh scattering) when the weak UV signal interacts with particles in free space [210]. In communication scenarios where UV LEDs are used as light sources, the constraints of the particle nature of the signal (i.e., wavelength limitations) on the one hand, and the process limitations of the analog modulation device on the other, are imposed. This makes it difficult for UV communication system to realize the precise modulation of the phase and frequency of the waveform as in the traditional RF communication, but rather the intensity modulation is realized in a number of code element cycles. Commonly used digital modulation techniques in UV communications include on-off keying (OOK), digital pulse interval modulation (DPIM), pulse position modulation (PPM), and a series of derivative versions of PPM modulation (this paper takes Digital Pulse Position Modulation as an example) [211], [212]. Methods such as m-pulse position modulation (MPPM) and quadrature amplitude modulation (QAM) modulation are also used in some high-speed communication scenarios.

For UV communication systems using intensity modulation, the intensity of the ambient light has a strong influence on the channel model. When the effect of ambient light is weak, the received signal is usually modeled using a Poisson process [213]. When the effect of ambient light on the signal is strong, the received signal is modeled as Additive White Gaussian Noise (AWGN) plus a DC offset [214]. In general, there are many measures of the performance of a digital communication system, the two most important of which are transmission efficiency and reliability [215]. Assume that the communication method is LOS communication, the channel model is Gaussian noise model, the number of source bits of the UV communication system is M, and the number of data bits transmitted in the channel is  $L_B = 2^M$ . When the bit rate  $R_{bit}$  and the total transmit power  $P_{total}$  are certain, the average transmit power, bandwidth, transmission capacity, and anticode interference performance of the above five modulation methods are shown in Table 5 [216], [217], [218], [219].

Let the total power  $P_{total}$  be 20mW, then the relationship between the average transmit power and the number of data bits is shown in Fig. 11(d). It can be found that the OOK modulation method has the maximum average transmit power and the PPM modulation method has the minimum average transmit power within M gaps when the number of bits of the transmitted signal is the same. Setting the bit rate  $R_{bit}$  to 6400bit/s, the curve of bandwidth versus the number of data bits is shown in Fig. 11(a). When the number of data bits increases, the bandwidth required for UV communication also increases, and when the amount of data is greater than 6 bits, the bandwidth required by the PPM modulation method is the largest and the bandwidth required by the OOK modulation method is the smallest. The transmission



FIGURE 11. Performance of different valence modulation techniques. (a) Relationship curve between average transmit power and number of data bits. (b) Relationship curve between bandwidth and number of data bits. (c) Relationship curve between transmission capacity and number of data bits. (d) Relationship curve between error time slot rate and signal-to-noise ratio.

capacity versus number of bits curve is shown in Fig. 11(c), from which it can be seen that the two have a linear or quasilinear relationship, and the transmission capacity of DPPM modulation is the largest, and the transmission capacity of PPM modulation is the smallest.

Assuming that the mean of the noise is 0 and the variance is  $\sigma_n^2$ , the formula for calculating the error time slot rate shown in Table 6 can be obtained when the optimal threshold values are obtained for the above four modulations [220]. where  $SNR = S_t/2\sigma_n^2$ . Setting M = 4, i.e., the number of symbol bits is  $L_B = 2^M$ , the relationship curve between the error time slot rate and the SNR is obtained, as shown in Fig. 11(b). From Fig. 11(d), it can be seen that the error time slot rate of all the four modulation modes decreases with the increase of the SNR, and when the SNR are the same, the error time slot rate of the PPM and OOK is larger than that of the DPPM and OOK modulation. In UV communication systems, the choice of modulation scheme significantly affects system performance. Under the same SNR conditions, DPPM and OOK modulation schemes have lower error time slot rates compared to PPM and OOK modulation schemes, which means that DPPM and OOK modulation schemes can provide more stable communication performance under the same SNR conditions. Additionally, in UV communication systems, increasing the SNR can effectively reduce the error time slot rate, thereby improving the performance of the communication system. Therefore, in practical applications, without changing

#### TABLE 5. Performance of four modulation methods

Modulation Method	OOK	DPIM	PPM	DPPM
Average Transmit Power	$P_{OOK} = \frac{P_{\text{total}}}{2}$	$P_{DPIM} = \frac{2P_{\text{total}}}{2M+3}$	$P_{PPM} = \frac{P_{\text{total}}}{2^M}$	$P_{DPPM} = \frac{2P_{\text{total}}}{2M+1}$
Bandwidth	$B_{OOK} = R_{\rm bit}$	$B_{DPIM} = \frac{(L_B+3)R_{\text{bit}}}{2\log_2 L_B}$ $= \frac{(2^M+3)R_{\text{bit}}}{2M}$	$B_{PPM} = \frac{L_B R_{\text{bit}}}{\log_2 L_B}$ $= \frac{2^M R_{\text{bit}}}{M}$	$B_{DPPM} = \frac{(L_B+1)R_{\text{bit}}}{2\log_2 L_B}$ $= \frac{(2^M+1)R_{\text{bit}}}{2M}$
Transmission Capacity	$C_{OOK} = M$	$C_{DPIM} = \frac{2M(2^{M}+1)}{2^{M}+3}$	$C_{PPM} = M$	$C_{DPPM}=2M2^M(2^M+1)$
Anti-Interference Ability	stronger	relatively weak	relatively weak	relatively strong

TABLE 6. Performance indicators (err	ror time slot rate) of four	modulation methods
--------------------------------------	-----------------------------	--------------------

Modulation Method	Missed Time Slot Rate
OOK	$\frac{1}{2}erfc\left[\frac{\sqrt{SNR}}{2}\right]$
DPIM	$\frac{1 + erfc\left[\frac{1}{2\sqrt{SNR}}ln\left(2^{M} + \frac{1}{2}\right) - \frac{1}{2}\sqrt{SNR}\right] + \frac{2^{M}+1}{2}\left[1 - erfc\left(\frac{1}{2\sqrt{SNR}}ln\left(2^{M} + \frac{1}{2} + \frac{1}{2}\sqrt{SNR}\right)\right)\right]}{2^{M}+3}$
PPM	$\frac{1 + erfc\left[\frac{1}{2\sqrt{SNR}}ln\left(2^{M}-1\right)-\frac{1}{2}\sqrt{SNR}\right] + (2^{M}-1)\left[1 - erfc\left(\frac{1}{2\sqrt{SNR}}ln\left(2^{M}-1\right)+\frac{1}{2}\sqrt{SNR}\right)\right]}{2^{M-1}}$
DPPM	$\frac{1 + erfc\left[\frac{(M-1)ln2}{2\sqrt{SNR}} - \frac{1}{2}\sqrt{SNR}\right] + \frac{2M-1}{2}\left[1 - erfc\left(\frac{(M-1)ln2}{2\sqrt{SNR}} + \frac{1}{2}\sqrt{SNR}\right)\right]}{2^{M} + 1}$

TABLE 7. Performance indicators of recently reported LOS UVC modulation technology.

Time	Modulation Method	Transmit Power	Link Length	Data Rate	Bit Error Rate	References
2019	OOK	196uW	0.3m	800Mbps	$3.8 \times 10^{-3}$	[226]
2019	OFDM	196uW	0.3m	1.1Gbps	$3.8 \times 10^{-3}$	[226]
2020	QAM	6mW	5m	2.4Gbps	—	[227]
2021	16-QAM-OFDM	0.854mW	0.5m	2Gbps	$3.8 \times 10^{-3}$	[228]
2021	16-QAM-OFDM	0.854mW	5m	0.82Gbps	$3.8 \times 10^{-3}$	[228]
2021	OFDM	2mW	5m	4Gbps	$3.8 \times 10^{-3}$	[229]
2022	QAM	0.4mW	17m	4Gbps	$3.8 \times 10^{-3}$	[230]
2022	QAM	0.567mW	0.5m	2.6Gbps	$3.8 \times 10^{-3}$	[231]
2022	OOK	3mW	2m	1.27Gbps	$2.0 \times 10^{-2}$	[232]
2022	QAM	0.4mW	10m	6.94Gbps	$3.8 \times 10^{-3}$	[233]
2022	QAM	0.4mW	116m	1.19Gbps	$3.8 \times 10^{-3}$	[233]

hardware conditions, the SNR can be improved by increasing transmission power, adopting more effective channel coding techniques, or modulation and demodulation techniques, to achieve better communication performance.

Table 7 shows the recently reported performance specifications of UV modulation techniques for UAV-oriented communication. In addition to the above modulation methods, QAM modulation is also a more widely used higher-order modulation scheme that can support Gbps communication rates. Compared with RF communication, UV communication has limited options for signal detection in medium and long range scenarios, most of which are directly detected, and suitable modulation and demodulation can improve the BER performance of the system to a certain extent [221].

Taking PPM modulation as an example, in the presence of strong interference such as log-normal atmospheric turbulence, the unconditional BER for binary PPM modulation in an UV communication system is approximately:

$$P_e = \frac{1}{\sqrt{\pi}} \sum_{i=1}^{n} w_i \mathcal{Q}\left(\frac{exp\left(2\left(\sqrt{2}\sigma_k x_i + m_k\right)\right)}{Fexp\left(\sqrt{2}\sigma_k x_i + m_k\right) + K_n}\right), \quad (9)$$

where Q is the electronic charge, F is the noise figure of the detector, and  $m_k$  is the mean of  $ln(K_s)$ .  $K_n = \frac{2\sigma_{Th}^2}{\overline{Gq}} + 2FK_{Bg}$ , and  $\sigma_k^2 = ln(\sigma_N^2 + 1)$ .  $[w_i]_{i=1}^n$  and  $[x_i]_{i=1}^n$  are the weights and zeros of the nth-order Hermite polynomials, respectively.  $K_{Bg} = \eta \lambda \frac{T_s}{hc}$  represents the average number of photons produced by background radiation power  $P_{Bg}$  within a PPM time slot.  $\sigma_{Th}^2 = (2\kappa T_e T_{s_p p m} \frac{q}{R_b})$  is the equivalent thermal noise within the PPM time slot. The fluctuation of  $K_s$  is caused by atmospheric turbulence, and its ensemble average is given by [222]:

$$E[K_s] = exp\left(\frac{\sigma_k^2}{2} + m_k\right). \tag{10}$$



FIGURE 12. Trend of BER with flicker index in UV LOS communication systems.



FIGURE 13. Trend of BER with flicker index in UV NLOS communication systems.

For an M-PPM system, the upper bound of the BER,  $P_e^M$ , is given by:

$$P_e^M = \frac{M}{2\sqrt{\pi}} \sum_{i=1}^n w_i \mathcal{Q}\left(\frac{exp\left(2\left(\sqrt{2}\sigma_k x_i + m_k\right)\right)}{Fexp\left(\sqrt{2}\sigma_k x_i + m_k\right) + K_n}\right).$$
(11)

According to Eq.11, the trend of the BER of UV LOS communication systems with varying scintillation indices at different turbulence levels is illustrated in Fig. 12. Considering the UV NLOS link as a superposition of LOS links, the trend of the BER for UV NLOS communication systems with varying scintillation indices is shown in Fig. 13. In the simulation, the bit rate  $R_b$  is set to  $155 \times 10^{-6}$ , the load resistance  $R_L$  is 50, the temperature is 300K, the electronic charge Q is  $1.602 \times 10^{-19}$ , and the Boltzmann constant k is  $1.38 \times 10^{-23}$ .

Fig. 12 and Fig. 13 illustrate that, under the same scintillation index, as the turbulence average intensity increases, the system's BER significantly rises. Under the same average intensity, an increase in the turbulence scintillation index leads to a higher BER. Additionally, turbulence has a greater impact on the performance of UV LOS communication systems compared to UV NLOS communication systems.

In practical applications, it is necessary to select a modulation scheme that is suitable for specific environmental conditions. For long-distance UV communication systems, the impact of weak turbulence on BER performance must be considered when designing the modulation scheme. For systems operating in strong turbulence environments, PPM is a better choice due to its superior BER performance under the same environmental conditions, which will be explained in the following content. Additionally, the design of UV communication systems also needs to take into account the influence of link type (LOS or NLOS) on system performance. LOS links are significantly affected by turbulence, hence require special attention during design. In contrast, NLOS links are less sensitive to turbulence, allowing other performance indicators to be focused on during design.

As can be inferred from the previous analysis, for modulation schemes with higher information rates such as OOK and QAM, their BER performance is worse than that of M-PPM under the same environmental conditions. The following section conducts a simulation analysis of the BER performance of UV communication systems in LOS and NLOS links under different turbulence intensities and scintillation indices, and examines their impact on various modulation schemes.

In weak turbulence channels, the intensity distribution follows a log-normal distribution, and the BER of the UV communication system in the LOS link can be represented as:

$$P_{LOS} = \int_0^\infty f(I)P(I) \, dI,\tag{12}$$

where f(I) is the probability density function of the normal distribution, given by:

$$f(I) = \frac{1}{\sqrt{2\pi}\sigma_s X} e^{-\frac{lnX + \frac{\sigma_s^2}{2}}{2\sigma_s^2}}.$$
 (13)

Here,  $\sigma_s^2$  is the log-amplitude variance, and X is the normalized average scintillation coefficient. P(I) represents the BER of the UV communication system under the condition of no turbulence, and when the modulation scheme is OOK, PPM, PAM, and QAM, the expressions for P(I) are respectively:

$$\begin{cases}
P_{OOK} = Q\left(\frac{\sqrt{SNR_{LOS}}}{2}\right), \\
P_{PPM} = \frac{L_B}{2}Q\left(\frac{\sqrt{L_B \log_2 L_B SNR_{LOS}}}{8}\right), \\
P_{PAM} = \frac{2(L_B - 1)}{L_B \log_2 L_B}Q\left(\frac{\sqrt{\log_2 L_B SNR_{LOS}}}{2(L_B - 1)}\right), \\
P_{QAM} = \frac{2(L_B - 1)}{L_B \log_2 L_B}Q\left(\frac{\sqrt{\log_2 L_B SNR_{LOS}}}{8(L_B - 1)}\right)
\end{cases}$$
(14)

VOLUME 5, 2024

where SNR is the SNR for UV LOS communication, and  $L_B$  is the number of bits in the modulated signal. The system's BER is then:

$$\begin{cases} P_{OOK,LOS} = \int_0^\infty f(I) Q\left(\frac{\sqrt{SNR_{LOS}I}}{2}\right) dI, \\ P_{PPM,LOS} = \int_0^\infty f(I) \frac{L_B}{2} Q\left(\frac{\sqrt{L_B \log_2 L_B SNR_{LOS}I}}{2}\right) dI, \\ P_{PAM,LOS} = \int_0^\infty f(I) \frac{2(L_B-1)}{L_B \log_2 L_B} Q\left(\frac{\sqrt{\log_2 L_B SNR_{LOS}I}}{2(L_B-1)}\right) dI, \\ P_{QAM,LOS} = \int_0^\infty f(I) \frac{2(L_B-1)}{L_B \log_2 L_B} Q\left(\frac{\sqrt{\log_2 L_B SNR_{LOS}I}}{8(L_B-1)}\right) dI \end{cases}$$
(15)

Similarly, treating the UV NLOS communication link as a combination of two LOS links, its probability density function is obtained from the conditional probability formula based on Eq. (14), eventually taking a form similar to Eq. (15). Set the communication distance of the system to 200m, the data transmission rate R to 0.5Mbps, the transmitter FOV angle to  $20^{\circ}$ , the receiver FOV angle to  $30^{\circ}$ , the elevation angle of the transmitter and receiver to  $30^{\circ}$ , and the transmission power to 50mW to simulate the performance of the system. The BER trends of UV LOS and UV NLOS communications under different turbulence scintillation indices with varying SNR are illustrated in Fig. 14.

From Fig. 14, it can be observed that as the scintillation index increases, the performance of the system's BER gradually deteriorates. Under the same turbulent conditions, the BER performance of the LOS link is superior to that of the NLOS link. Under the same turbulent conditions and link structure, the BER performance of PPM modulation is better than that of OOK modulation, OOK modulation is better than that of PAM modulation, and QAM modulation has the worst performance. This indicates that for specific conditions, the optimal modulation scheme can be found, which satisfies both the rate and the BER. Therefore, in strong interference or long distance communication scenarios, other approaches such as adaptive modulation techniques are required to obtain better BER or higher communication rates. Specifically, selecting an appropriate modulation scheme requires consideration of the system's power efficiency and channel coding scheme. In some cases, adopting more complex modulation techniques may result in higher communication rates, but it will also increase system complexity and power consumption. Therefore, when designing a modulation scheme, it is necessary to combine methods such as swarm intelligence, deep learning, and convex optimization to achieve an effective trade-off between communication rate, BER performance, power efficiency, and system complexity.

For fading caused by atmospheric turbulence, the transmitter can change the transmission parameters such as power, modulation size, and code rate according to the channel conditions, i.e., adaptive coding and modulation (ACM). There are three common types of adaptive modulation: variable rate variable power systems, variable rate constant power systems and constant rate variable power systems [223]. Based on these classifications, it can be found that adaptive modulation technique is not a standalone technique, it is often combined with beamforming, power control and other techniques for integrated design, as related to the relevant sections in the previous section. For the modulation technique itself, [224] utilized the constant envelope property of S-PSK to adjust the modulation order of SPSK according to the turbulence-induced transient state and BER requirements, resulting in an effective increase in the capacity of the FSO channel. In order to improve the spectral efficiency, [225] changed the modulation order of the MPSK scheme according to the instantaneous channel state and BER limits to improve the spectral efficiency. Reference [137] by adopting N-ary coding adaptive spatial modulation (SM) in FSO communication and combining it with M-ary PPM, and using diversity methods with MIMO and MISO configurations, significantly alleviated the signal attenuation caused by atmospheric turbulence, thereby reducing the system's BER. Simulation results show that this approach enhances the stability and integrity of the signal, laying a foundation for high-capacity, robust optical wireless communication, and can effectively address the challenges posed by atmospheric turbulence. In addition to this, adaptive modulation techniques are mostly used as a comparison strategy in the design of HD systems and beamforming systems, and there are still fewer related researches specialized in adaptive modulation techniques for UV communication.

#### X. CODING

The UAV transmits a wide range of data types during its mission, including data information such as images and videos and control information such as position, attitude, altitude and speed. The structure of the transmission system includes air-to-air transmission and air-to-ground transmission. Similar to adaptive modulation techniques, adaptive coding techniques aim to trade-off rate and BER in different scenarios [234], [235]. Related studies have pointed out that when the transmission rate of the system is less than the channel capacity given by Shannon's theorem, an arbitrarily small BER can be obtained at a given SNR by appropriate coding error correction techniques [236]. For practical UV communication systems, under specific turbulence intensity and link distance conditions, the selection of appropriate coding techniques can significantly reduce the BER of the system or achieve the minimum transmit power at a given BER condition. The latter is important for reducing ambient radiation versus improving the energy utilization of UV communication systems for UAVs [237], [238].

In recent years, the more widely used channel coding techniques are Reed-Solomon (RS) codes, Low Density Parity Check (LDPC) codes, Polar codes and Turbo codes. The latter three error-correcting codes have attracted wide attention in RF communication due to their excellent error-correction performance and ability to approach the Shannon limit [239], [240], [241], and the performance advantages and disadvantages of the above error-correcting codes in



FIGURE 14. The relationship between BER and SNR in UV communication system under turbulent conditions: (a) LOS link,  $\sigma_s = 0.1$ ; (b) NLOS link,  $\sigma_s = 0.1$ ; (c) LOS link,  $\sigma_s = 0.3$ ; (d) NLOS link,  $\sigma_s = 0.3$ ; (e) LOS link,  $\sigma_s = 0.5$ ; (f) NLOS link,  $\sigma_s = 0.5$ .

different types of communication systems have become one of the focuses of attention. Reference [242] investigated the farthest communication link for UV communication supported by RS code and LDPC code under a given power and BER. The experimental results show that the communication distance supported by LDPC codes is more than twice that of RS codes, i.e., LDPC has better communication performance. Reference [243] compared the performance difference between LDPC codes and Turbo codes in UV communication systems and concluded that LDPC codes have better BER performance and complexity than Turbo codes. References [241] and [244] compared the BER and secrecy performance of Polar and LDPC codes, respectively. The authors point out that Polar codes have better BER performance and secrecy than LDPC codes. Polar codes, as a later born channel code, the improvement of its codeword construction and decoding method is an important research direction, while there are still fewer cases about the application of Turbo in UV communication. Therefore, the current research status of LDPC codes and Polar codes in UV communication systems in recent years will be described next.

In terms of the algorithmic application of LDPC, [245] applied LDPC codes to UV communication systems and

is found that the LDPC code helps a lot to improve the performance of the system, but this improvement is limited by the distance, and the BER shows a waterfall jump trend with the increase of distance. When the transmission distance varies around a certain threshold, the BER changes drastically. When the transmission distance is much larger than this threshold, the improvement of BER by LDPC code is not obvious, and when the transmission distance is smaller than this threshold, the improvement of BER by LDPC code is better. Reference [246] combined the LDPC coding scheme with spread spectrum technology to propose an effective system scheme. The results of analyzing the system model show that the scheme can effectively suppress multipath interference and improve the BER performance of UV communication. Reference [156], on the other hand, applied the LDPC algorithm to the coplanar MIMO system as a preamble channel coding algorithm for different merging algorithms in the diversity technique, and achieved reliable analytical results. Regarding the physical structure configuration of the LDPC system, taking the [247] as an example, the authors proposed a scheme of UV communication system that employs LDPC coding, and this study investigated the BER performance of

simulated the BER performance at different distances. It

the system by taking into account the macroscopic physical parameters of the device, the microscopic mechanism, and the system structural parameters. The study reveals the advantages of adaptive coding.

In terms of performance enhancement and analysis of LDPC compiled code algorithms, [248] proposed a heuristic checksum matrix construction algorithm for LDPC codes to approximate the upper bound of the data transmission rate based on the received response characteristics of the UV scattering channel. The algorithm achieves a better BER on short-link UV communication systems compared to conventional LDPC coding methods. References [246] and [249] investigated the performance of LDPC coding in UV communication systems at different code lengths and BER algorithms. The results show that the belief propagation (BP) algorithm with log-likelihood-ratio confidence propagation (LLR BP) has superior BER performance compared to the bit-flip (BF) decoding algorithm and the minimum-sum (MS) decoding algorithm. Beyond that, long codes should be used in UV communication systems. Reference [250] expanded the previous results of LDPC coding research under the single scattering assumption to multiple scattering channels, established a Monte Carlo simulation model, and analyzed the impact of different decoding algorithms on the performance of UV communication systems. The results show that the BP decoding algorithm has the best performance compared to BF and MS algorithms. From these studies, it can be found that the integration of LDPC codes with practical UV communication systems has been widely used, especially for SISO and LOS types of systems. So the extension of the application to 3D high-order MIMO systems is a feasible research direction, while the performance advantages of Turbo codes have not been fully released, the research focus will still be on the decoding algorithm.

Unlike LDPC codes and Turbo codes, Polar codes can be theoretically proved to reach the capacity of arbitrary binary input discrete memoryless channels. But due to its late start, its code word construction and decoding algorithms are not as mature as those of LDPC codes and Turbo codes, and its application in UV communication systems is not as widespread as that of LDPC codes. At present, the performance enhancement of Polar codes under finite length is a key research direction in this field. Reference [251] optimized the successive cancellation decoding algorithm by using the path loss model of UV communication, and implemented the successive cancellation list decoding algorithm. The results show that the SCL algorithm can improve the effective communication distance of the UV communication system, and the performance of the SCL decoding algorithm increases with the increase of SC. In order to optimize the bottleneck of performance enhancement that occurs after the number of SC increases to a certain degree, the article introduced Cyclic Redundancy Check (CRC) so that the system obtains a lower BER. Reference [252] considered the computational complexity issue from the perspective of coding length, derived the

VOLUME 5, 2024

upper limit of Bhattacharyya for general polar codes in UV communication systems, and then give the length of the minimum codeword. The coding scheme for Polar codes proposed in the article can obtain better performance with lower complexity. In addition to this, coding methods applied to UAV swarm UV communication have also been proposed by scholars, such as [253], which proposes unequal level coding Lu by Transform (ULC-LT) and stepwise unequal error protection Lu by Transform (SUEP-LT) based on the fountain code, and the results show that both coding methods have a better performance than LT coding.

In order to trade-off the reliability and throughput of UV communication, we expect the transmitter to be able to adjust the parameters, which has been proved to be effective in RF communication systems [254], [255]. In the existing research on VLC communication systems, various types of adaptive transmission systems have been proposed and their feasibility and advantages have been verified. Safi provided an adaptive coding and power control scheme and analyzes the effect of channel estimation error in practical FSO systems [256]. Reference [257], on the other hand, investigated the joint adaptation of modulation size and transmission power. The study proposed an adaptive noncoding transmission algorithm that enables the transmission power and modulation size of the system to vary with channel conditions. Reference [258] investigated the coding system with fixed code rate and MPPM under average power constraints, and proposed an adaptive coding scheme for Turbo codes, which enables the system to select the modulation size at the transmitter based on CSI. References [259] and [260] designed adaptive coding rate adjustment algorithms based on LDPC codes, which are capable of achieving optimal throughput based on the real-time characteristics of the channel.

To address the negative impact of channel conditions on the system, [261] investigated the joint adaption of symbol rate and power, and the article considered adaptive symbol rate with symbol duration varying with channel conditions as a way to reduce the effect of atmospheric scintillation in FSO links. In order to mitigate the performance degradation in the presence of strong turbulence, [262] investigated three LDPC coded modulation schemes, namely, variable-rate variablepower adaptive scheme, fixed-rate channel inversion scheme, and truncated channel inversion scheme. The article pointed out that the fixed-rate channel inversion scheme is suitable for weak turbulence conditions, and the variable rate variable power scheme or the truncated channel inversion scheme is suitable for strong turbulence situations.

This chapter presents a study of coding techniques in UV communications, beginning with an introduction to commonly used LDPC and Turbo codes and their decoding algorithms, followed by a consideration of adaptive problems in the coding domain. As with techniques such as beamforming, power optimization, and confidentiality optimization, such adaptive problems are often modeled as optimization problems dependent on the channel state. In the field of UAV UV communication, adaptive coding techniques are a more flexible direction than traditional research on coding algorithms [263], [264]. For example, incorporating both adaptive coding and adaptive modulation techniques in systems that aim to reduce inter-link interference. Or in a HD system, use a solution to adjust the encoding bit rate instead of changing the modulation depth. The prerequisite for these techniques lies in the utilization of real-time characteristics of the UV channel. Not only that, but as a basis for implementing adaptive algorithms with lower time complexity, we would like to model non-black-box forms of more concise but no less accurate optimization in these cross-directions.

# **XI. CONCLUSION AND FUTURE DIRECTIONS**

This paper provides a comprehensive overview of research on UV communication technology for UAVs. We present eight aspects of the current state of research in UV communications for UAVs from a communications engineering perspective: channel modeling, localization techniques, beamforming, power control, networking, diversity and merging, modulation and demodulation, and coding. With the help of this survey, readers can learn about the current state of development of UV communication technology for UAVs, so that they can conduct further research on the technical details of interest.

In the preceding analysis, the article was structured according to various technical categories. In dynamic environments, changes in external environmental parameters necessitate that the UV communication system automatically adapts to the impact on system parameters. This adaptive mechanism must be capable of integrating with multiple technologies to comprehensively enhance system performance. The implementation of adaptive technologies in UV communication systems primarily encompasses three aspects: adaptive coding, adaptive modulation, and beamforming.

In the realm of adaptive coding, the system is capable of selecting appropriate coding schemes based on the realtime characteristics of the channel conditions. For instance, when the channel conditions are favorable, the system may opt for LDPC codes to increase data transmission rates. Conversely, when channel conditions deteriorate, the system may switch to Turbo codes to bolster interference resilience. Furthermore, the system can dynamically adjust the coding rate; for example, it may increase the coding rate (e.g., from 1/2 to 3/4) under high signal quality conditions, while reducing the coding rate (e.g., from 3/4 to 1/2) when signal quality declines, thereby ensuring communication reliability.

Regarding adaptive modulation, the system can select suitable modulation schemes based on the instantaneous state of the channel and the requirements for BER. For example, under good channel conditions, the system may employ 64-QAM (Quadrature Amplitude Modulation) to achieve higher data transmission rates, whereas it may resort to 16-QAM or Quadrature Phase Shift Keying (QPSK) under poorer channel conditions to enhance signal robustness against interference. Additionally, by jointly adjusting modulation size and transmission power, the system can optimize performance across varying channel conditions; for instance, it may increase transmission power to extend range when signal quality is high, while reducing transmission power to mitigate interference when signal quality is low.

Beamforming technology optimizes the propagation direction of signals by utilizing directional antenna arrays. In multi-user environments, the system can employ adaptive beamforming techniques to focus signals towards specific users, thereby enhancing the signal strength for those users while reducing interference to others. Simultaneously, the system can dynamically adjust the amplitude and direction of the beam based on real-time channel states and user positions, ensuring stability and efficiency in signal transmission.

The comprehensive application of these adaptive technologies can significantly enhance the communication efficiency and reliability of UV communication systems. In addition to this, based on the previous research work, we summarize the future development trend of UAV UV communication technology:

- Channel modeling: Firstly, to address the challenges of communication systems in special contexts, including the effects of UAV flight characteristics, airframe vibration, altitude change pairs, and special reflective surfaces and obstacles on communication, further refinement of channel models is needed to improve communication quality. Secondly, as the transmission distance increases, the difference in average path loss between high-order scattering and second-order scattering is approximately 15dB, highlighting the increased significance of multiple scattering effects. Trends indicated in Fig. 4 suggest that future research will focus on developing models capable of accurately considering multiple scattering to improve the precision of UV channels. Lastly, it is crucial to prioritize the computational efficiency of the model (reducing the time complexity from  $O(n^4)$  or even higher to  $O(n^2)$  or O(n), enabling the system to obtain channel information for UAV-based UV communication in realtime and efficiently.
- Positioning techniques: Firstly, for three-dimensional spatial positioning techniques, existing research has proposed methods that achieve average positioning accuracy of 3.5cm and 6cm, respectively. Studies indicate that determining height based on two-dimensional localization can achieve higher precision in positioning. Secondly, since the transmitter in actual UV communication systems can point in any direction, researching positioning techniques based on NLOS links has become an important direction. Additionally, combining the advantages of LOS links and NLOS links, the focus of future research will be on multi-link fusion positioning techniques. Currently proposed methods can

achieve an accuracy rate of over 80% with an average error of less than 10m within a space of 200m in length, width, and height. Lastly, the application of UV image-based positioning techniques on mobile platforms is of significant importance. Current technology has achieved stable communication connections within a horizontal FOV of 320 degrees and a vertical FOV of 110 degrees. Therefore, the future focus of UV communication positioning technology research will be on three-dimensional spatial positioning techniques, NLOS link positioning techniques, multi-link fusion positioning techniques, UV image-based positioning techniques, and multidimensional data fusion positioning techniques. The development of these directions will promote the efficient, safe, and reliable application of UV communication in UAV groups.

- Beamforming Technology: The objective of UV communication beamforming technology is to minimize mutual interference in MIMO and network communication scenarios while enhancing the confidentiality of signal transmission. According to previous research, different input signal distributions have varying impacts on the secrecy rate, with truncated Gaussian distribution input signals showing better performance in terms of secrecy rate. Therefore, it is necessary to adjust the distribution characteristics of the signals to further enhance the system's confidentiality. Secondly, in multi-user MISO communication scenarios, the design of effective precoding schemes can further improve the system's secrecy and anti-interference capabilities. Current research has been able to significantly enhance the system's secrecy and fairness through precoding schemes designed based on optimization models of upper and lower bounds. Additionally, studies have shown that addressing the issue of target channel and eavesdropper channel position changes in UAV communication scenarios, by analyzing the random distribution of eavesdroppers, can effectively improve system security. By minimizing the average SNR of the eavesdropper and maximizing the SNR of the target user, the system's security can be significantly improved. Therefore, the future focus of research on UV communication beamforming technology will be on enhancing the confidentiality of the signal transmission process, designing effective precoding schemes, and analyzing the issue of target channel and eavesdropper channel position changes in UAV communication scenarios, with actual data supporting these research directions.
- Power optimization: Compared to UV communication systems without power optimization technology, those that employ power optimization techniques can effectively reduce environmental radiation. For UV-specific power optimization systems, they can significantly decrease the system's transmission power

while maintaining the BER, providing a solid foundation for outdoor applications of UAV-based UV communication. The development direction of power optimization technology for UAV-based UV communication mainly includes three aspects. Firstly, by integrating the characteristics of UV channels, establishing a power control model suitable for UV communication systems to enhance communication quality and energy efficiency. Secondly, researching hybrid dimming schemes that utilize the advantages of both HD and SD to expand the dynamic range of power and enhance system robustness. Lastly, improving the execution efficiency of optimization algorithms, using the information from the model itself to provide a basis for execution, in order to enhance the accuracy of the optimization model and the efficiency of the algorithm.

- Diversity techniques: First, further optimization of diversity reception schemes, such as the use of multibeam methods to improve system throughput, the combination of appropriate MIMO coding to achieve higher diversity gain, and the improvement of coding schemes such as STC, DSTBC, and so on, in order to reduce the system BER. The second is to explore the performance of different combining methods under different modulation conditions, such as comparing the performance of SC, EGC and MRC combining methods under BPSK modulation, as well as exploring the combination of new modulation methods and combining schemes, such as the combination of SOPP-OSTBC and 2PPM modulation. Future research focus will be extended to MIMO systems above second order to further refine the application of diversity techniques in UV communications.
- Networking technology: The development direction of UV networking technology mainly focuses on dynamic TDMA protocols, contention-based random access protocols, and optimized network topology and node density. As shown in Fig. 7, dynamic TDMA protocols improve network performance through dynamic time slot allocation mechanisms, while contention-based random access protocols allow nodes to compete for channel access during idle time slots. The former achieves higher throughput, lower latency, and lower packet loss rates through dynamic time slot allocation mechanisms, while the latter can effectively reduce interference between signals, thereby improving transmission efficiency and the overall data throughput of the network. In addition, optimizing network topology and node density is crucial for improving network connectivity and throughput. Through improvements in these three areas, UV networking technology can better adapt to changes in network topology and node roles, enhancing network flexibility and reliability. At the same time, for special application scenarios shown in Fig. 6, research on aspects such as UAV

swarm collaboration, obstacle avoidance, and relay node optimization should be strengthened to further improve the performance and communication range of UV communication networks.

- Modulation technology: Based on the results shown in Fig. 9, higher-order modulation schemes can increase channel capacity and transmission rates, but they also reduce the system's anti-interference capability and BER performance under complex conditions. In Fig. 12, with relatively deteriorated channel conditions, when the turbulence intensity and scintillation index are the same, OAM modulation has the highest BER, while PPM modulation has the lowest BER. As the SNR decreases, the BER magnitude between the two increases gradually from about  $10^{-6}$  dB to close to 1 dB. Therefore, the following aspects can be considered for the improvement of modulation schemes. First, integrating techniques such as beamforming and power control to achieve a comprehensive design and improve system adaptability and performance. Second, utilizing modulation schemes with constant envelope characteristics, such as S-PSK, and adjusting them according to channel conditions and BER requirements to increase communication capacity. Third, adopting more flexible modulation schemes, such as MPSK, based on instantaneous channel conditions and BER constraints to improve spectral efficiency. The development of these technologies will enable UAV UV communication systems to better adapt to different environments and communication requirements, enhancing the reliability and efficiency of communication.
- Coding technology: The introduction of this paper on UV communication coding technology points out that LDPC codes have received considerable attention due to their excellent error correction performance and ability to approach the Shannon limit, while the application of Turbo codes in UV communication is not yet widespread. Therefore, related research will continue to explore how to improve system performance by adjusting coding techniques, especially in the optimization and application under specific transmission distances and environmental conditions. Secondly, as an emerging coding technology, the performance enhancement of Polar codes at finite lengths is the focus of current research, and the optimization and application of their decoding algorithms will be the main research direction in the future. Finally, with the development of artificial intelligence and machine learning, adaptive coding technology for UAV UV communication may introduce more intelligent adaptive algorithms, improving the flexibility and adaptability of the system, thus further enhancing communication efficiency and performance. These development trends will make the coding technology of UAV UV communication play a more important role in the future and provide higherquality communication services for various application scenarios.

Unlike the well-known RF, IR and VLC communications, UV communication technology is relatively less well known. UAV-based UV communication has been less researched, and the challenges it faces (availability, stability, effectiveness) are more severe. However, the various properties of UV communication have shown its potential to be applied to UAV communication (For example, as indicated in Fig. 8, UV communication has the advantage of low path loss in short-distance NLOS scenarios.), and more research work will be conducted in the future centering on the above challenges.

#### REFERENCES

- B. Fan, Y. Li, R. Zhang, and Q. Fu, "Review on the technological development and application of UAV systems," *Chin. J. Electron.*, vol. 29, no. 2, pp. 199–207, 2020.
- [2] R. B. Yesilay and A. Macit, "Economic outlook of unmanned aerial vehicles in turkey and the world: Drone economies'," in *Proc. 3rd Int. Conf. Econ. Res.*, vol. 24, 2019, p. 209.
- [3] M. Li, Y. Hong, C. Zeng, Y. Song, and X. Zhang, "Investigation on the UAV-to-satellite optical communication systems," *IEEE J. Sel. Areas Commun*, vol. 36, no. 9, pp. 2128–2138, Sep. 2018.
- [4] W. Fawaz, C. Abou-Rjeily, and C. Assi, "UAV-aided cooperation for FSO communication systems," *IEEE Commun Mag.*, vol. 56, no. 1, pp. 70–75, Jan. 2018.
- [5] Y. Jiang, X. Li, G. Zhu, H. Li, J. Deng, and Q. Shi, "6G nonterrestrial networks enabled low-altitude economy: Opportunities and challenges," 2023, arXiv:2311.09047.
- [6] Z. Xu and B. M. Sadler, "Ultraviolet communications: Potential and state-of-the-art," *IEEE Commun Mag.*, vol. 46, no. 5, pp. 67–73, May 2008.
- [7] H. Yu et al., "Dual-functional triangular-shape micro-size lightemitting and detecting diode for on-chip optical communication in the deep ultraviolet band," *Laser Photon. Rev.*, vol. 18, no. 4, 2024, Art. no. 2300789.
- [8] S. Li, Z. Liu, M. Zhang, L. Yang, Y. Guo, and W. Tang, "Highperformance self-powered GaN/PEDOT: PSS hybrid heterojunction UV photodetector for optical communication," *Sci. China Technol. Sci.*, vol. 67, no. 2, pp. 608–615, 2024.
- [9] Q. Tang et al., "Enhanced photoresponse of self-powered ZnObased photoelectrochemical-type UV photodetectors via ga-doping for optical communication application," J. Alloys Compd., vol. 972, Jan. 2024, Art. no. 172864.
- [10] Z. Ying, W. Feigang, H. Hua, S. Peng, and K. Xizheng, "Research progress of ultraviolet non-direct communication," *Study Opt. Commun.*, vol. 1, no. 2, p. 1, 2024.
- [11] W. Chen, J. Cheng, C. Wu, L. Chen, and H. Zhang, "BCH based U-UV codes and its SCL decoding," *IEEE Trans. Signal Process.*, vol. 72, pp. 1286–1300, Feb. 2024.
- [12] L. Wang et al., "All-light communication network for spaceair-sea integrated interconnection," *Opt. Express*, vol. 32, no. 6, pp. 9219–9226, 2024.
- [13] G. A. Shaw, M. L. Nischan, M. A. Iyengar, S. Kaushik, and M. K. Griffin, "NLOS UV communication for distributed sensor systems," in *Proc. SPIE Integr. Command Environ.*, 2000, pp. 83–96.
- [14] G. Chen, T. Wu, F. Yang, T. Wang, J. Song, and Z. Han, "Ultravioletbased UAV swarm communications: Potentials and challenges," *IEEE Wireless Commun.*, vol. 29, no. 5, pp. 84–90, Oct. 2022.
- [15] K. K. Garg, P. Shaik, R. Sharma, P. Brida, O. Krejcar, and V. Bhatia, "On the capacity of a SIM-based cooperative NLOS UVC system with best relay selection," *ICT Express*, vol. 10, no. 2, pp. 285–291, 2024.
- [16] A. Vavoulas, H. G. Sandalidis, N. D. Chatzidiamantis, Z. Xu, and G. K. Karagiannidis, "A survey on ultraviolet C-band (UV-C) communications," *IEEE Commun. Surveys Tuts.*, vol. 21, no. 3, pp. 2111–2133, 3rd Quart., 2019.
- [17] R. Yuan and J. Ma, "Review of ultraviolet non-line-of-sight communication," *China Commun*, vol. 13, no. 6, pp. 63–75, Jun. 2016.

- [18] G. A. Shaw, A. M. Siegel, J. Model, and D. Greisokh, "Recent progress in short-range ultraviolet communication," in *Proc. 7th SPIE* Unattended Ground Sens. Technol. Appl., 2005, pp. 214–225.
- [19] S. Arya and Y. H. Chung, "A comprehensive survey on optical scattering communications: Current research, new trends, and future vision," *IEEE Comms. Surveys Tut.*, vol. 26, no. 2, pp. 1446–1477, 2nd Quart., 2023.
- [20] M. R. Hayal et al., "Modeling and investigation on the performance enhancement of hovering UAV-based FSO relay optical wireless communication systems under pointing errors and atmospheric turbulence effects," *Opt. Quantum Electron.*, vol. 55, no. 7, p. 625, 2023.
- [21] B. L. Diffey, "Sources and measurement of ultraviolet radiation," *Methods*, vol. 28, no. 1, pp. 4–13, 2002.
- [22] J. D'Orazio, S. Jarrett, A. Amaro-Ortiz, and T. Scott, "UV radiation and the skin," *Int. J. Mol. Sci.*, vol. 14, no. 6, pp. 12222–12248, 2013.
- [23] X. Tang, T. Yang, D. Yu, H. Xiong, and S. Zhang, "Current insights and future perspectives of ultraviolet radiation (UV) exposure: Friends and foes to the skin and beyond the skin," *Environ. Int.*, vol. 185, Mar. 2024, Art. no. 108535.
- [24] M.-H. Shen and R. K. Singh, "Elicitation and disinfection during sprout production using ultraviolet radiation and hydrogen peroxide: A review," *Trends Food Sci. Technol.*, vol. 147, May 2024, Art. no. 104447.
- [25] M. R. Luettgen, J. H. Shapiro, and D. M. Reilly, "Non-line-of-sight single-scatter propagation model," *J. Opt. Soc. Amer. A*, vol. 8, no. 12, pp. 1964–1972, 1991.
- [26] A. Du et al., "Implementation of large angle low error non-line-ofsight ultraviolet communication system," *Opt. Commun.*, vol. 558, May 2024, Art. no. 130227.
- [27] X. Ke, "Ultraviolet non-line-of-sight communication," in *Handbook of Optical Wireless Communication*. Singapore: Springer, 2024, pp. 1619–1646.
- [28] K. Yue, "Application of EEUC-based inter aircraft ultraviolet communication network algorithm in energy consumption optimization of drone swarm," *Energy Inform.*, vol. 7, no. 1, p. 27, 2024.
- [29] Y. Sun, Y. Zuo, T. Luo, Z. Chen, and Z. Zhang, "Hybrid space-time coding technique in UV communication system," in *Proc. 3rd SPIE Int. Conf. Electron. Inf. Eng., Big Data, Comput. Technol. (EIBDCT)*, 2024, pp. 19–23.
- [30] D. M. Reilly and C. Warde, "Temporal characteristics of singlescatter radiation," J. Opt. Soc. Amer., vol. 69, no. 3, pp. 464–470, 1979.
- [31] Z. Xu, H. Ding, B. M. Sadler, and G. Chen, "Analytical performance study of solar blind non-line-of-sight ultraviolet short-range communication links," *Opt. Lett.*, vol. 33, no. 16, pp. 1860–1862, 2008.
- [32] G. Chen, Z. Xu, H. Ding, and B. M. Sadler, "Path loss modeling and performance trade-off study for short-range non-line-of-sight ultraviolet communications," *Opt. Express*, vol. 17, no. 5, pp. 3929–3940, 2009.
- [33] M. A. Elshimy and S. Hranilovic, "Impact of finite receiver-aperture size in a non-line-of-sight single-scatter propagation model," J. Opt. Soc. Amer. A, vol. 28, no. 12, pp. 2568–2576, 2011.
- [34] L. Wang, Z. Xu, and B. M. Sadler, "An approximate closed-form link loss model for non-line-of-sight ultraviolet communication in noncoplanar geometry," *Opt. Lett.*, vol. 36, no. 7, pp. 1224–1226, 2011.
- [35] Y. Zuo, H. Xiao, J. Wu, W. Li, and J. Lin, "Closed-form path loss model of non-line-of-sight ultraviolet single-scatter propagation," *Opt. Lett.*, vol. 38, no. 12, pp. 2116–2118, 2013.
- [36] H. Ding, G. Chen, A. K. Majumdar, B. M. Sadler, and Z. Xu, "Modeling of non-line-of-sight ultraviolet scattering channels for communication," *IEEE J. Sel. Areas Commun.*, vol. 27, no. 9, pp. 1535–1544, Dec. 2009.
- [37] R. J. Drost, T. J. Moore, and B. M. Sadler, "UV communications channel modeling incorporating multiple scattering interactions," J. Opt. Soc. Amer. A, vol. 28, no. 4, pp. 686–695, 2011.
- [38] D. Han, X. Fan, K. Zhang, and R. Zhu, "Research on multiplescattering channel with monte carlo model in UV atmosphere communication," *Appl. Opt.*, vol. 52, no. 22, pp. 5516–5522, 2013.
- [39] W. Liu, D. Zou, and Z. Xu, "Modeling of optical wireless scattering communication channels over broad spectra," *J. Opt. Soc. Amer. A*, vol. 32, no. 3, pp. 486–490, 2015.

- [40] C. Xu and H. Zhang, "Channel analyses over wide optical spectra for long-range scattering communication," *IEEE Commun. Lett.*, vol. 19, no. 2, pp. 187–190, Feb. 2015.
- [41] C. Xu, H. Zhang, and J. Cheng, "Effects of haze particles and fog droplets on NLOS ultraviolet communication channels," *Opt. Expess*, vol. 23, no. 18, pp. 23259–23269, 2015.
- [42] R. J. Drost, B. M. Sadler, and G. Chen, "Dead time effects in non-line-of-sight ultraviolet communications," *Opt. Express*, vol. 23, no. 12, pp. 15748–15761, 2015.
- [43] C. Xu and H. Zhang, "Monte-carlo based modeling for ultraviolet non-line-of-sight communication channels with typical obstacles," in *Proc. Int. Conf. Wireless Commun. Signal Process. (WCSP)*, 2015, pp. 1–5.
- [44] P. Song, X. Zhou, F. Song, C. Su, and A. Wang, "Performance analysis of UV multiple-scatter communication system with height difference," *Appl. Opt.*, vol. 56, no. 32, pp. 8908–8916, 2017.
- [45] T. Wu, J. Ma, P. Su, R. Yuan, and J. Cheng, "Modeling of short-range ultraviolet communication channel based on spherical coordinate system," *IEEE Commun. Lett.*, vol. 23, no. 2, pp. 242–245, Feb. 2019.
- [46] Z. Shen, J. Ma, T. Shan, and T. Wu, "Modeling of ultraviolet scattering propagation and its applicability analysis," *Opt. Lett.*, vol. 44, no. 20, pp. 4953–4956, 2019.
- [47] T. Wu, J. Ma, R. Yuan, P. Su, and J. Cheng, "Single-scatter model for short-range ultraviolet communication in a narrow beam case," *IEEE Photon. Technol. Lett.*, vol. 31, no. 3, pp. 265–268, Feb. 1, 2019.
- [48] R. Yuan, J. Ma, P. Su, and Z. He, "An integral model of two-order and three-order scattering for non-line-of-sight ultraviolet communication in a narrow beam case," *IEEE Commun. Lett.*, vol. 20, no. 12, pp. 2366–2369, Dec. 2016.
- [49] T. Shan, J. Ma, T. Wu, Z. Shen, and P. Su, "Single scattering turbulence model based on the division of effective scattering volume for ultraviolet communication," *Chin. Opt. Lett.*, vol. 18, no. 12, 2020, Art. no. 120602.
- [50] T. Shan, J. Ma, T. Wu, Z. Shen, and P. Su, "Modeling of ultraviolet omni-directional multiple scattering channel based on monte carlo method," *Opt. Lett.*, vol. 45, no. 20, pp. 5724–5727, 2020.
- [51] R. Yuan, J. Ma, P. Su, Y. Dong, and J. Cheng, "Monte-carlo integration models for multiple scattering based optical wireless communication," *IEEE Trans. Commun.*, vol. 68, no. 1, pp. 334–348, Jan. 2020.
- [52] Z. Shen, J. Ma, T. Shan, and P. Su, "Improved monte carlo integration models for ultraviolet communications," in *Proc. IEEE Int. Conf. Commun.Technol. (ICCT)*, 2020, pp. 168–172.
- [53] Z. Shen, J. Ma, S. B. Provost, and P. Su, "Effects of transceiver jitter on the performance of optical scattering communication systems," *Opt. Lett.*, vol. 45, no. 20, pp. 5680–5683, 2020.
- [54] Y. Ma and J. Zhou, "Effects of mixed aerosol on the path loss of NLOS UV communication system," *Optoelectron. Lett.*, vol. 20, no. 9, pp. 549–559, 2024.
- [55] C. H. Arslan, F. T. Dagefu, T. J. Moore, M. J. Weisman, and R. J. Drost, "Measurement system for ultraviolet channel modeling and communications," *Opt. Express*, vol. 31, no. 15, pp. 23714–23728, 2023.
- [56] B. Li, H. Wang, M. Liu, H. Hu, and Z. Mao, "Applicability of nonline-of-sight ultraviolet single-scatter approximation model," *Photon. Netw. Commun.*, vol. 31, pp. 147–154, Feb. 2016.
- [57] J. Hu et al., "A brief review on the positioning technologies for unmanned aerial vehicles," in *Proc. IEEE Int. Conf. Unmanned Syst.* (*ICUS*), 2017, pp. 527–532.
- [58] W. Zhang and M. Kavehrad, "Comparison of VLC-based indoor positioning techniques," in *Proc. 7th SPIE Broadband Access Commun. Technol.*, 2013, pp. 152–157.
- [59] J. Armstrong, Y. A. Sekercioglu, and A. Neild, "Visible light positioning: A roadmap for international standardization," *IEEE Commun Mag.*, vol. 51, no. 12, pp. 68–73, Dec. 2013.
- [60] T. Q. Wang, Y. A. Sekercioglu, A. Neild, and J. Armstrong, "Position accuracy of time-of-arrival based ranging using visible light with application in indoor localization systems," *J. Lightw. Technol.*, vol. 31, no. 20, pp. 3302–3308, Oct. 15, 2013.
- [61] D. Niculescu and B. Nath, "Ad Hoc positioning system (APS) using AOA," in Proc. 22nd Annu. Joint Conf. IEEE Comput. Commun. Soc., vol. 3, 2003, pp. 1734–1743.

- [62] A. Nasipuri and K. Li, "A directionality based location discovery scheme for wireless sensor networks," in *Proc. 1st ACM Int. Workshop Wireless Sens. Netw. Appl.*, 2002, pp. 105–111.
- [63] S.-Y. Jung, S. Hann, S. Park, and C.-S. Park, "Optical wireless indoor positioning system using light emitting diode ceiling lights," *Microw. Opt. Techn. Lett.*, vol. 54, no. 7, pp. 1622–1626, 2012.
- [64] Z. Zhou, M. Kavehrad, and P. Deng, "Indoor positioning algorithm using light-emitting diode visible light communications," *Opt. Eng.*, vol. 51, no. 8, 2012, Art. no. 85009.
- [65] H.-S. Kim, D.-R. Kim, S.-H. Yang, Y.-H. Son, and S.-K. Han, "An indoor visible light communication positioning system using a RF carrier allocation technique," *J. Lightw. Technol.*, vol. 31, no. 1, pp. 134–144, Jan. 1, 2013.
- [66] Q.-L. Li, J.-Y. Wang, T. Huang, and Y. Wang, "Three-dimensional indoor visible light positioning system with a single transmitter and a single tilted receiver," *Opt. Eng.*, vol. 55, no. 10, 2016, Art. no. 106103.
- [67] S.-H. Yang, H.-S. Kim, Y.-H. Son, and S.-K. Han, "Threedimensional visible light indoor localization using AOA and RSS with multiple optical receivers," *J. Lightw. Technol.*, vol. 32, no. 14, pp. 2480–2485, Jul. 15, 2014.
- [68] Y. Wang, Y. Gong, and Z. Shi, "Research on the collinear equation model of visual positioning based on visible light communication," in *Proc. MATEC Web Conf.*, vol. 22, 2015, p. 2003.
- [69] W. Gu, M. Kavehrad, and M. Aminikashani, "Three-dimensional indoor light positioning algorithm based on nonlinear estimation," in *Proc. 10th SPIE Broadband Access Commun. Technol.*, 2016, pp. 201–206.
- [70] L. Yin, X. Wu, and H. Haas, "Indoor visible light positioning with angle diversity transmitter," in *Proc. IEEE 82nd Veh. Technol. Conf.*, 2015, pp. 1–5.
- [71] Y. Xu, J. Zhao, J. Shi, and N. Chi, "Reversed three-dimensional visible light indoor positioning utilizing annular receivers with multiphotodiodes," *Sensors*, vol. 16, no. 8, p. 1254, 2016.
- [72] G. Zhang and L.-T. Hsu, "Intelligent GNSS/INS integrated navigation system for a commercial UAV flight control system," *Aerosp. Sci. Technol.*, vol. 80, pp. 368–380, 2018.
- [73] H. Du et al., "Demonstration of a pulsed ultraviolet laser positioning system based on 4H-SiC four-quadrant photodetectors," *IEEE Photon. Technol. Lett.*, vol. 36, no. 11, pp. 697–700, Jun. 1, 2024.
- [74] C. H. Arslan, F. T. Dagefu, J. N. Twigg, and T. J. Moore, "Nonline-of-sight ranging via ultraviolet signals," in *Proc. Int. Conf. Mil. Commun. Inf. Syst. (ICMCIS)*, 2023, pp. 1–6.
- [75] R. Yuan, S. Wang, G. Liu, and M. Peng, "Non-line-of-sight ultraviolet positioning using linearly-arrayed photon-counting receivers," *IEEE J. Sel. Areas Commun.*, vol. 42, no. 9, pp. 2535–2549, Sep. 2024.
- [76] S. Wang et al., "Non-line-of-sight ultraviolet positioning using two photon-counting receivers," in *Proc. IEEE Global Commun. Conf.*, 2023, pp. 3682–3687.
- [77] H. Qi, C. Gong, and Z. Xu, "Omnidirectional antenna array-based transmitter direction sensing in ultra-violet Ad-Hoc scattering communication networks," in *Proc. IEEE Int. Conf. Commun. Workshops* (ICC Workshops), 2019, pp. 1–6.
- [78] T. Zhao, P. Liu, S. Zhang, and Q. Ma, "UAV assisted landing guided by UV LEDs," *Appl. Opt.*, vol. 59, no. 20, pp. 5910–5917, 2020.
- [79] H. Shi, T. Zhao, and H. Zhang, "Wireless ultra violet light positioning and guidance system during UAV flight," in *Proc. SPIE Int. Conf. Optoelectron. Microelectron. Technol. Appl.*, 2020, pp. 181–191.
- [80] T. Zhao, X. Yu, P. Liu, and L. Liu, "Ultraviolet anti-collision and localization algorithm in UAV formation network," *Optik*, vol. 192, Sep. 2019, Art. no. 162919.
- [81] V. Walter, M. Saska, and A. Franchi, "Fast mutual relative localization of UAVs using ultraviolet led markers," in *Proc. Int. Conf. Unmanned Aircr. Syst. (ICUAS)*, 2018, pp. 1217–1226.
- [82] V. Walter, N. Staub, M. Saska, and A. Franchi, "Mutual localization of uavs based on blinking ultraviolet markers and 3D time-position hough transform," in *Proc. IEEE 14th Int. Conf. Autom. Sci. Eng.* (*CASE*), 2018, pp. 298–303.
- [83] V. Walter, N. Staub, A. Franchi, and M. Saska, "UVDAR system for visual relative localization with application to leader–follower formations of multirotor UAVs," *IEEE Robot. Autom.*, vol. 4, no. 3, pp. 2637–2644, Jul. 2019.

- [84] J. Horyna, V. Walter, and M. Saska, "UVDAR-COM: UV-based relative localization of UAVs with integrated optical communication," in *Proc. Int. Conf. Unmanned Aircr. Syst. (ICUAS)*, 2022, pp. 1302–1308.
- [85] N. Huang, F. Chen, C. Liu, Y.-Y. Zhang, Y.-J. Zhu, and C. Gong, "Robust design and experimental demonstration of RSS-based visible light positioning under limited receiving FOV," *IEEE Trans. Green Commun. Netw.*, early access, Aug. 13, 2024, doi: 10.1109/TGCN.2024.3443086.
- [86] J. De Bruycker et al., "An experimental testbed for the performance evaluation of optical time difference of arrival based indoor positioning," *J. Lightw. Technol.*, early access, Jul. 10, 2024, doi: 10.1109/JLT.2024.3426033.
- [87] H. Yang et al., "An advanced integrated visible light communication and localization system," *IEEE Trans. Commun.*, vol. 71, no. 12, pp. 7149–7162, Dec. 2023.
- [88] A. Harlakin and P. A. Hoeher, "LCD-based angle-of-arrival estimation of multiple light sources: Paving the road to multiuser VLC," *IEEE Photon. J.*, vol. 16, no. 3, pp. 1–15, Jun. 2024.
- [89] M. Feng, Y. Wang, M. Li, S. Liu, G. Huang, and P. Li, "Design of OCC indoor positioning system based on flat panel light and angle sensor assistance," *Appl. Sci.*, vol. 13, no. 8, p. 4745, 2023.
- [90] C. Isaia and M. P. Michaelides, "A review of wireless positioning techniques and technologies: From smart sensors to 6G," *Signals*, vol. 4, no. 1, pp. 90–136, 2023.
- [91] J. Bi et al., "PSOSVRPos: WiFi indoor positioning using SVR optimized by PSO," *Expert Syst. Appl.*, vol. 222, Jul. 2023, Art. no. 119778.
- [92] X. Liu, X. Sun, and X. Xia, "LiDAR point's elliptical error model and laser positioning for autonomous vehicles," *Meas. Sci. Technol.*, vol. 32, no. 3, 2020, Art. no. 35107.
- [93] Q. Xu, C. Jiang, Y. Han, B. Wang, and K. R. Liu, "Waveforming: An overview with beamforming," *IEEE Commun. Surveys Tuts.*, vol. 20, no. 1, pp. 132–149, 1st Quart., 2017.
- [94] A. Mostafa and L. Lampe, "Physical-layer security for MISO visible light communication channels," *IEEE J. Sel. Areas Commun.*, vol. 33, no. 9, pp. 1806–1818, Sep. 2015.
- [95] S. Hranilovic and F. Kschischang, "Optical intensity-modulated direct detection channels: Signal space and lattice codes," *IEEE Trans. Inf. Theory.*, vol. 49, no. 6, pp. 1385–1399, Jun. 2003.
- [96] A. Lapidoth, S. M. Moser, and M. A. Wigger, "On the capacity of free-space optical intensity channels," *IEEE Trans. Inf. Theory.*, vol. 55, no. 10, pp. 4449–4461, Oct. 2009.
- [97] T. Y. Aung, S. Arya, and Y. H. Chung, "A dynamic beamforming technique for ultraviolet-based indoor communications," *IEEE Sensors J.*, vol. 20, no. 18, pp. 10547–10553, Sep. 2020.
- [98] S. Arya and Y. H. Chung, "State-of-the-art ultraviolet multiuser indoor communication over power-constrained discrete-time poisson channels," *Opt. Eng.*, vol. 59, no. 10, 2020, Art. no. 106106.
- [99] S. A. Al-Gailani et al., "A survey of free space optics (FSO) communication systems, links, and networks," *IEEE Access*, vol. 9, pp. 7353–7373, 2020.
- [100] A. D. Wyner, "The wire-tap channel," *Bell Syst. Tech. J.*, vol. 54, no. 8, pp. 1355–1387, Oct. 1975.
- [101] A. Mostafa and L. Lampe, "Pattern synthesis of massive LED arrays for secure visible light communication links," in *Proc. IEEE Int. Conf. Commun. Workshop (ICCW)*, 2015, pp. 1350–1355.
- [102] A. Mostafa and L. Lampe, "Optimal and robust beamforming for secure transmission in MISO visible-light communication links," *IEEE Trans. Signal Process.*, vol. 64, no. 24, pp. 6501–6516, Dec. 2016.
- [103] H. Zaid, Z. Rezki, A. Chaaban, and M. S. Alouini, "Improved achievable secrecy rate of visible light communication with cooperative jamming," in *Proc. IEEE Global Conf. Signal Inf. Process.* (*GlobalSIP*), 2015, pp. 1165–1169.
- [104] M. A. Arfaoui, Z. Rezki, A. Ghrayeb, and M. S. Alouini, "On the secrecy capacity of MISO visible light communication channels," in *Proc. IEEE Global Commun. Conf. (GLOBECOM)*, 2016, pp. 1–7.
- [105] T. V. Pham, H. Le-Minh, and A. T. Pham, "Multi-user visible light communication broadcast channels with zero-forcing precoding," *IEEE Trans. Commun.*, vol. 65, no. 6, pp. 2509–2521, Jun. 2017.

- [106] T. V. Pham, T. Hayashi, and A. T. Pham, "Artificial-noise-aided precoding design for multi-user visible light communication channels," *IEEE Access*, vol. 7, pp. 3767–3777, 2019.
- [107] M. A. Arfaoui, A. Ghrayeb, and C. M. Assi, "Secrecy performance of multi-user MISO VLC broadcast channels with confidential messages," *IEEE Trans. Wireless Commun.*, vol. 17, no. 11, pp. 7789–7800, Nov. 2018.
- [108] S. Cho, G. Chen, and J. P. Coon, "Secrecy analysis in visible light communication systems with randomly located eavesdroppers," in *Proc. IEEE Int. Conf. Commun. Workshops (ICC Workshops)*, 2017, pp. 475–480.
- [109] S. Cho, G. Chen, and J. P. Coon, "Securing visible light communication systems by beamforming in the presence of randomly distributed eavesdroppers," *IEEE Trans. Wireless Commun.*, vol. 17, no. 5, pp. 2918–2931, May 2018.
- [110] S. Cho, G. Chen, and J. P. Coon, "Physical layer security in visible light communication systems with randomly located colluding eavesdroppers," *IEEE Wireless Commun. Lett.*, vol. 7, no. 5, pp. 768–771, Oct. 2018.
- [111] A. Kumar and S. K. Ghorai, "High-gain and volume-efficient concave-convex lens for multiple-input multiple-output visible light communication system," *Opt. Eng.*, vol. 59, no. 3, 2020, Art. no. 35106.
- [112] Z. Wang, R. Yuan, J. Cheng, and M. Peng, "Joint optimization of full-duplex relay placement and transmit power for multi-hop ultraviolet communications," *IEEE Internet Things J.*, vol. 11, no. 7, pp. 11960–11973, Apr. 2024.
- [113] P. D. Diamantoulakis, G. K. Karagiannidis, and Z. Ding, "Simultaneous lightwave information and power transfer (SLIPT)," *IEEE Trans. Green Commun.*, vol. 2, no. 3, pp. 764–773, Sep. 2018.
- [114] C. Gong, S. Li, Q. Gao, and Z. Xu, "Power and rate optimization for visible light communication system with lighting constraints," *IEEE Trans. Signal Process.*, vol. 63, no. 16, pp. 4245–4256, Aug. 2015.
- [115] X. Liu, A. Yang, Y. Li, and L. Feng, "Separate dimming controlling and data transmission for an indoor visible light communication system," *China Commun*, vol. 12, no. 3, pp. 71–76, Mar. 2015.
- [116] S. Ma, R. Yang, H. Li, Z.-L. Dong, H. Gu, and S. Li, "Achievable rate with closed-form for SISO channel and broadcast channel in visible light communication networks," *J. Lightw. Technol.*, vol. 35, no. 14, pp. 2778–2787, Jul. 15, 2017.
- [117] S. Ma, F. Zhang, H. Li, F. Zhou, Y. Wang, and S. Li, "Simultaneous lightwave information and power transfer in visible light communication systems," *IEEE Trans. Wireless Commun.*, vol. 18, no. 12, pp. 5818–5830, Dec. 2019.
- [118] D. Tagliaferri, M. Brambilla, M. Nicoli, and U. Spagnolini, "Sensoraided beamwidth and power control for next generation vehicular communications," *IEEE Access*, vol. 9, pp. 56301–56317, 2021.
- [119] I. Din and H. Kim, "Energy-efficient brightness control and data transmission for visible light communication," *IEEE Photon. Technol. Lett.*, vol. 26, no. 8, pp. 781–784, Apr. 2014.
- [120] K. Lee and H. Park, "Modulations for visible light communications with dimming control," *IEEE Photon. Technol. Lett.*, vol. 23, no. 16, pp. 1136–1138, Aug. 2011.
- [121] A. B. Siddique and M. Tahir, "Joint rate-brightness control using variable rate MPPM for LED based visible light communication systems," *IEEE Trans. Wireless Commun.*, vol. 12, no. 9, pp. 4604–4611, Sep. 2013.
- [122] H. Sugiyama, S. Haruyama, and M. Nakagawa, "Brightness control methods for illumination and visible-light communication systems," in *Proc. 3rd Int. Conf. Wireless Mobile Commun.* (*ICWMC*), 2007, pp. 78–78.
- [123] Z. Wang, W.-D. Zhong, C. Yu, J. Chen, C. P. S. Francois, and W. Chen, "Performance of dimming control scheme in visible light communication system," *Opt. Express*, vol. 20, no. 17, pp. 18861–18868, 2012.
- [124] M. T. Alresheedi and J. M. Elmirghani, "Mobile optical wireless systems employing beam angle and power adaptation with diversity receivers," in *Proc. 7th Int. Conf. Wireless Opt. Commun. Netw.* (WOCN), 2010, pp. 1–6.
- [125] H. Ji, T. Zhang, S. Qiao, and Z. Ghassemlooy, "Joint dimming control and optimal power allocation for THO-OFDM visible light communications," *IEEE Trans. Commun.*, vol. 69, no. 8, pp. 5352–5366, Aug. 2021.

- [126] J.-Y. Wang, J.-B. Wang, N. Huang, and M. Chen, "Capacity analysis for pulse amplitude modulated visible light communications with dimming control," *J. Opt. Soc. Amer. A*, vol. 31, no. 3, pp. 561–568, 2014.
- [127] Y. Yang, C. Wang, C. Feng, C. Guo, J. Cheng, and Z. Zeng, "A generalized dimming control scheme for visible light communications," *IEEE Trans. Commun.*, vol. 69, no. 3, pp. 1845–1857, Mar. 2021.
- [128] Y. Yang, Z. Zeng, J. Cheng, and C. Guo, "A novel hybrid dimming control scheme for visible light communications," *IEEE Photon. J.*, vol. 9, no. 6, pp. 1–12, Dec. 2017.
- [129] T. Wu, F. Yang, J. Song, J. Ma, and P. Su, "Modeling of UV diffused-LOS communication channel incorporating obstacle and its applicability analysis," *Opt. Lett.*, vol. 46, no. 18, pp. 4578–4581, 2021.
- [130] L. Liao, "Long distance non-line-of-sight ultraviolet communication channel analysis and experimental verification," Ph.D. dissertation, Dept. Electr. Eng., UC Riverside, Riverside, CA, USA, 2015.
- [131] D. Han, M. Zhang, and Q. Li, "Experimental measurement of link gain and correlation in a single-input multiple-output ultraviolet communication system with diversity reception," *Opt. Eng.*, vol. 56, no. 8, 2017, Art. no. 84108.
- [132] X. Zhu and J. M. Kahn, "Free-space optical communication through atmospheric turbulence channels," *IEEE Trans. Commun.*, vol. 50, no. 8, pp. 1293–1300, Aug. 2002.
- [133] L. C. Andrews, R. L. Phillips, and C. Y. Hopen, *Laser Beam Scintillation with Applications*, vol. 99, Bellingham, WA, USA: SPIE Press, 2001.
- [134] U. H. Rizvi, F. Yilmaz, M.-S. Alouini, G. J. M. Janssen, and J. H. Weber, "Performance of equal gain combining with quantized phases in rayleigh fading channels," *IEEE Trans. Commun.*, vol. 59, no. 1, pp. 13–18, Jan. 2010.
- [135] S. H. Hariq and N. Odabasioglu, "Spatial diversity techniques for non-line-of-sight ultraviolet communication systems over atmospheric turbulence channels," *IET Optoelectron*, vol. 14, no. 5, pp. 327–336, 2020.
- [136] R. Yuan and M. Peng, "Single-input multiple-output scattering based optical communications using statical combining in turbulent channels," *IEEE Trans. Wireless Commun.*, vol. 23, no. 4, pp. 2560–2574, Apr. 2024.
- [137] E. E. Elsayed, "Atmospheric turbulence mitigation of MIMO-RF/FSO DWDM communication systems using advanced diversity multiplexing with hybrid N-SM/OMI M-ary spatial pulse-position modulation schemes," *Opt. Commun.*, vol. 562, Jul. 2024, Art. no. 130558.
- [138] E. J. Lee and V. W. S. Chan, "Part 1: Optical communication over the clear turbulent atmospheric channel using diversity," *IEEE J. Sel. Areas Commun.*, vol. 22, no. 9, pp. 1896–1906, Nov. 2004.
- [139] D. Han, Y. Liu, K. Zhang, P. Luo, and M. Zhang, "Theoretical and experimental research on diversity reception technology in NLOS UV communication system," *Opt. Exp.*, vol. 20, no. 14, pp. 15833–15842, 2012.
- [140] L. Guo, D. Meng, K. Liu, X. Mu, W. Feng, and D. Han, "Experimental research on the MRC diversity reception algorithm for UV communication," *Appl. Opt.*, vol. 54, no. 16, pp. 5050–5056, 2015.
- [141] X. Meng, M. Zhang, D. Han, L. Song, and P. Luo, "Experimental study on 1× 4 real-time SIMO diversity reception scheme for a ultraviolet communication system," in *Proc. 20th Eur. Conf. Netw. Opt. Commun. (NOC)*, 2015, pp. 1–4.
- [142] J. Shi et al., "Experimental study of the mobility feature and selective maximal ratio combining algorithm for UV communication," in *Proc.* 20th Eur. Conf. Netw. Opt. Commun. (NOC), 2015, pp. 1–4.
- [143] X. Wang, M. Zhang, D. Han, X. Wang, and R. Zhu, "Implementation and experimental demonstration of multiple receptions in solar-blind UV communications," in *Proc. Asia Commun. Photon. Conf. (ACP)*, 2013, Art. no. AF2K-6.
- [144] K. K. Garg and V. Bhatia, "Performance analysis of cooperative NLOS UVC system with receiver diversity," in *Proc. Nat. Conf. Commun. (NCC)*, 2020, pp. 1–6.
- [145] H. Xiao, Y. Zuo, J. Wu, Y. Li, and J. Lin, "Bit-error-rate performance of non-line-of-sight UV transmission with spatial diversity reception," *Opt. Lett.*, vol. 37, no. 19, pp. 4143–4145, 2012.

- [146] D. Tiwari, S. Soni, and P. S. Chauhan, "A new closed-form expressions of channel capacity with MRC, EGC and SC over lognormal fading channel," *Wireless Pers. Commun.*, vol. 97, no. 3, pp. 4183–4197, 2017.
- [147] Y. Zuo, H. Xiao, H. Qin, Y. Zhao, X. Zhang, and J. Wu, "Performance of non-line-of-sight ultraviolet spatial diversity reception in turbulence," in *Proc. Asia Commun. Photon. Conf. (ACP)*, 2013, Art. no. AF2F-72.
- [148] S. Arya and Y. H. Chung, "M-PSK subcarrier intensity modulation with switch-and-stay diversity for NLOS ultraviolet communication," in *Proc. IEEE Region 10 Conf. (TENCON)*, 2018, pp. 524–529.
- [149] W. Mou, T. Pu, W. Yang, J. Zheng, and X. Tang, "Secrecy performance of noncoplanar SIMOME NLOS ultraviolet communications over turbulence channels," *Opt. Commun.*, vol. 433, pp. 262–267, Feb. 2019.
- [150] A. Gupta and M. Brandt-Pearce, "Receiver design for shot noise limited MIMO FSO/UV communication systems," in *Proc. IEEE Globecom Workshops*, 2012, pp. 1183–1187.
- [151] M. Biagi, A. M. Vegni, S. Pergoloni, P. M. Butala, and T. D. C. Little, "Trace-orthogonal PPM-space time block coding under rate constraints for visible light communication," *J. Lightw. Technol.*, vol. 33, no. 2, pp. 481–494, Jan. 2015.
- [152] L. Guo, K. Liu, D. Meng, X. Mu, and D. Han, "Simulation and experimental research on the Alamouti code for ultraviolet communication," *Opt. Eng.*, vol. 55, no. 1, 2016, Art. no. 15101.
- [153] Y. M. Khattabi and M. M. Matalgah, "Alamouti-OSTBC wireless cooperative networks with mobile nodes and imperfect CSI estimation," *IEEE Trans. Veh. Technol.*, vol. 67, no. 4, pp. 3447–3456, Apr. 2018.
- [154] L. Guo, X. Mu, X. Pan, K. Liu, D. Meng, and D. Han, "DSTBC experimental research on UV communication system," *Photonic Netw. Commun.*, vol. 33, pp. 69–76, Feb. 2017.
- [155] Y. Gu, M. Zhang, D. Han, Q. Chen, and Z. Ghassemlooy, "Experimental research on SOPP-OSTBC scheme in UV communication with concise 2-PPM," in *Proc. Conf. Lasers Electro-Opt. Pac. Rim (CLEO-PR)*, 2017, pp. 1–3.
- [156] J. Ji, P. Su, J. Ma, B. Xia, and X. Xue, "LDPC channel coding in coplanar MIMO ultraviolet communication system," in *Proc. 20th Int. Conf. Opt. Commun. Netw. (ICOCN)*, 2022, pp. 1–3.
- [157] O. Aboelala, I. E. Lee, and G. C. Chung, "A survey of hybrid free space optics (FSO) communication networks to achieve 5G connectivity for backhauling," *Entropy*, vol. 24, no. 11, p. 1573, 2022.
- [158] D. M. Reilly, D. T. Moriarty, and J. A. Maynard, "Unique properties of solar blind ultraviolet communication systems for unattended ground sensor networks," in *Unmanned/Unattended Sensors and Sensor Networks*. Bellingham, WA, USA: SPIE, 2004, pp. 244–254.
- [159] M. J. Weisman, F. T. Dagefu, T. J. Moore, C. H. Arslan, and R. J. Drost, "Analysis of the low-probability-of-detection characteristics of ultraviolet communications," *Opt. Exp.*, vol. 28, no. 16, pp. 23640–23651, 2020.
- [160] M. A. Karabulut, A. F. M. S. Shah, and H. Ilhan, "A novel MIMO-OFDM based MAC protocol for VANETs," *IEEE Trans. Intell. Transp. Syst.*, vol. 23, no. 11, pp. 20255–20267, Nov. 2022.
- [161] J.-B. Mao, J.-H. Li, J.-Y. Wang, and W. Wei, "A novel ultraviolet communication channel access protocol based on competition mechanism," *Optik*, vol. 273, Feb. 2023, Art. no. 170426.
- [162] Y. Li, J. Ning, Z. Xu, S. V. Krishnamurthy, and G. Chen, "UVOC-MAC: A MAC protocol for outdoor ultraviolet networks," *Wireless Netw.*, vol. 19, pp. 1101–1120, Aug. 2013.
- [163] Y. Pan, G. Wang, Y. Zhang, J. Tang, C. Gong, and Z. Xu, "Graph-based conflict-free MAC protocol and conflict analysis for a two-layer ultraviolet communication network," *J. Opt. Commun. Netw.*, vol. 15, no. 6, pp. 381–392, Jun. 2023.
- [164] C. Li et al., "Enhanced clustering MAC protocol based on learning automata for UV networks," *Photonics*, vol. 11, no. 4, p. 340, 2024.
- [165] S. Johari and M. B. Krishna, "TDMA based contention-free MAC protocols for vehicular ad hoc networks: A survey," *Veh. Commun.*, vol. 28, Apr. 2021, Art. no. 100308.
- [166] J. Zhang, W. Liu, N. Huang, and Z. Xu, "A hybrid MAC protocol with dynamic resource allocation for underwater optical wireless networks," in *Proc. IEEE 11th Int. Conf. Inf., Commun. Netw.* (*ICICN*), 2023, pp. 37–42.

- [167] X.-Z. Ke, H. He, and C.-I. Wu, "A new ant colony-based routing algorithm with unidirectional link in UV mesh communication wireless network," *Optoelectron Lett.*, vol. 7, no. 2, pp. 139–142, 2011.
- [168] T. Zhao, Y. Gao, P. Wu, Y. Xie, and P. Song, "A networking strategy for three-dimensional wireless ultraviolet communication network," *Optik*, vol. 151, pp. 123–135, Dec. 2017.
- [169] Z. Gao, D. Han, M. Zhang, X. Jiang, C. Yang, and L. Wang, "A multipoint access scheme for ultraviolet wireless network," in *Proc. IEEE* 6th Int. Conf. Comput. Commun. (ICCC), 2020, pp. 1302–1306.
- [170] H. Qi, D. Zou, C. Gong, and Z. Xu, "Two-dimensional intensity distribution and adaptive power allocation for ultraviolet ad-hoc network," *IEEE Trans. Green Commun. Netw.*, vol. 6, no. 1, pp. 558–570, Mar. 2022.
- [171] Z. Taifei, G. Yingying, and Z. Ying, "An area coverage algorithm for non-line-of-sight ultraviolet communication network," *Photonic Netw. Commun.*, vol. 32, pp. 269–280, Mar. 2016.
- [172] H. Qin, Y. Zuo, F. Li, R. Cong, L. Meng, and J. Wu, "Noncoplanar geometry for mobile NLOS MIMO ultraviolet communication with linear complexity signal detection," *IEEE Photon. J.*, vol. 9, no. 5, pp. 1–12, Oct. 2017.
- [173] X. Ke and Y. Wu, "Diamond communication network design for the non-line-of-sight ultraviolet channel model," *Opt. Eng.*, vol. 60, no. 3, 2021, Art. no. 36104.
- [174] F. Cao, C. Li, and Z. Xu, "Research on the coverage of ultraviolet communication network in the arbitrary polygon area," *Photonics*, vol. 8, no. 7, p. 240, 2021.
- [175] T. Zhao, Y. Xie, and Y. Zhang, "Connectivity properties for UAVs networks in wireless ultraviolet communication," *Photonic Netw Commun.*, vol. 35, pp. 316–324, Jan. 2018.
- [176] P. von Rickenbach, R. Wattenhofer, and A. Zollinger, "Algorithmic models of interference in wireless ad hoc and sensor networks," *IEEE* ACM Trans Netw., vol. 17, no. 1, pp. 172–185, Feb. 2009.
- [177] R. K. Ganti and M. Haenggi, "Interference and outage in clustered wireless ad hoc networks," *IEEE Trans. Inf. Theory.*, vol. 55, no. 9, pp. 4067–4086, Sep. 2009.
- [178] Z. Taifei and S. Shiyuan, "A fault-tolerant topology control algorithm for wireless ultraviolet light covert communication network in cluster UAV," *Photonic Netw. Commun.*, vol. 42, no. 1, pp. 40–48, 2021.
- [179] T. Zhao, X. Lv, K. Rong, and H. Zhang, "Optimization target variable ultraviolet optical communication network fault tolerant topology control method," in *Proc. Int. Conf. Neural Netw., Inf. Commun. Eng.*, 2021, pp. 348–353.
- [180] T. Zhao, Q. Li, and P. Song, "A fast channel assignment scheme based on power control in wireless ultraviolet networks," *Comput. Electr. Eng.*, vol. 56, pp. 262–276, Nov. 2016.
- [181] T. Zhao, J. Yao, C. Gong, and Y. Wang, "Wireless ultraviolet light MIMO assisted UAV direction perception and collision avoidance method," *Phys. Commun.*, vol. 54, Oct. 2022, Art. no. 101815.
- [182] C. Li, J. Li, Z. Xu, and J. Wang, "Study on the k-connectivity of ultraviolet communication network under uniform distribution of nodes in a circular region," in *Proc. IEEE 5th Int. Conf. Comput. Commun. (ICCC)*, 2019, pp. 797–802.
- [183] H. Tadayyoni and M. Uysal, "Ultraviolet communications for groundto-air links," in *Proc. 27th Signal Process. Commun. Appl. Conf.* (SIU), 2019, pp. 1–4.
- [184] H. Tadayyoni, M. H. Ardakani, A. R. Heidarpour, and M. Uysal, "Ultraviolet communications for unmanned aerial vehicle networks," *IEEE Wireless Commun. Lett.*, vol. 11, no. 1, pp. 178–182, Jan. 2022.
- [185] C. Prehofer and C. Bettstetter, "Self-organization in communication networks: Principles and design paradigms," *IEEE Commun. Mag.*, vol. 43, no. 7, pp. 78–85, Jul. 2005.
- [186] S. Pandey, P. Shukla, and A. Tripathi, "An efficient group-based neighbor discovery for wireless sensor networks," in *Proc. 3rd Int. Conf. Intell. Commun. Technol. Virtual Mobile Netw. (ICICV)*, 2021, pp. 173–180.
- [187] G. Pei, M. Albuquerque, J. H. Kim, D. P. Nast, and P. R. Norris, "A neighbor discovery protocol for directional antenna networks," in *Proc. IEEE Mil. Commun. Conf. (MILCOM)*, 2005, pp. 487–492.
- [188] J. Luo and D. Guo, "Neighbor discovery in wireless ad hoc networks based on group testing," in *Proc. 46th Annu. Allerton Conf. Commun., Control, Comput.*, 2008, pp. 791–797.

- [189] M. J. McGlynn and S. A. Borbash, "Birthday protocols for low energy deployment and flexible neighbor discovery in ad hoc wireless networks," in *Proc. 2nd ACM Int. Symp. Mobile Ad Hoc Netw. Comput.*, 2001, pp. 137–145.
- [190] G. Jakllari, W. Luo, and S. V. Krishnamurthy, "An integrated neighbor discovery and MAC protocol for ad hoc networks using directional antennas," *IEEE Trans. Wireless Commun.*, vol. 6, no. 3, pp. 1114–1024, Mar. 2007.
- [191] S. Vasudevan, D. Towsley, D. Goeckel, and R. Khalili, "Neighbor discovery in wireless networks and the coupon collector's problem," in *Proc. 15th Annu. Int. Conf. Mobile Comput. Netw.*, 2009, pp. 181–192.
- [192] T. Yin, Y. Wang, M. Zhao, and J. Xiao, "A modified dynamic TDMA slot allocation algorithm in ad hoc network," in *Proc. 1st IEEE Int. Conf. Comput. Commun. Internet (ICCCI)*, 2016, pp. 124–128.
- [193] C. Li, Z. Xu, J. Wang, J. Zhao, A. Qi, and J. Li, "Ultraviolet random access collaborative networking protocol based on time division multiple access," *J. Opt. Commun. Netw.*, vol. 15, no. 6, pp. 393–403, Jun. 2023.
- [194] M. Li, T. Liu, and Y. Yang, "A low delay dynamic distributed TDMA slot allocation algorithm for ad hoc networks," in *Proc. Int. Conf. Intell. Syst., Commun., Comput. Netw. (ISCCN)*, 2023, pp. 489–494.
- [195] Y. Yu, L. Ru, K. Fang, and X. Jia, "Comprehensive mobility prediction based clustering algorithm for ad hoc UAV networks," in *Proc. 4th Int. Conf. Mach., Mater. Comput. Technol.*, 2016, pp. 600–613.
- [196] Y. Zeng and R. Zhang, "Energy-efficient UAV communication with trajectory optimization," *IEEE Trans. Wireless Commun.*, vol. 16, no. 6, pp. 3747–3760, Jun. 2017.
- [197] W. Wu, N. Xiong, and C. Wu, "Improved clustering algorithm based on energy consumption in wireless sensor networks," *IET Netw.*, vol. 6, no. 3, pp. 47–53, 2017.
- [198] J. Yang, J. Luo, F. Cheng, and D. Fu, "A stability routing protocol for the solar blind ultraviolet ad-hoc network," in *Proc. Int. Conf. Intell. Syst. Res. Mechatronics Eng.*, 2015, pp. 926–930.
- [199] L. Wang, Y. Li, Z. Xu, and S. V. Krishnamurthy, "A novel neighbor discovery protocol for ultraviolet wireless networks," in *Proc. 14th ACM Int. Conf. Model., Anal. Simul. Wireless Mobile Syst.*, 2011, pp. 135–142.
- [200] Y. Gang, L. Xiaoyi, C. Mou, and T. Mian, "A new timedivision multiple access Neighbor discovery algorithm for ultraviolet self-Organizing networks," *Optoelectron. Laser*, vol. 26, no. 6, pp. 1074–1080, 2015.
- [201] Z. Xiwon, Z. Shanghong, L. Yongjun, D. Boyu, and C. Zhen, "Directional MAC protocol for multi-channel inter-vehicle ultraviolet communication based on spatial division multiplexing," *Laser Technol.*, vol. 40, no. 3, pp. 451–455, 2016.
- [202] Y. Li, L. Wang, Z. Xu, and S. V. Krishnamurthy, "Neighbor discovery for ultraviolet ad hoc networks," *IEEE J. Sel. Areas Commun.*, vol. 29, no. 10, pp. 2002–2011, Dec. 2011.
- [203] C. Li, J. Li, Z. Xu, and J. Wang, "Study on the k-connectivity of UV communication network under the node distribution of RWP mobility model in the arbitrary polygon area," *IEEE Photon. J.*, vol. 12, no. 4, pp. 1–12, Aug. 2020.
- [204] T. Zhao, Y. Xie, S. Xu, J. Wang, and H. Wang, "Flocking of UAV formation with wireless ultraviolet communication," *Wireless Pers. Commun.*, vol. 114, pp. 2551–2568, May 2020.
- [205] K. Liu, T. Zhao, and P. Gao, "Wireless UV light UAV inter-aircraft avoidance algorithm based on target trajectory prediction," in *Proc. Int. Conf. Neural Netw., Inf. Commun. Eng.*, 2021, pp. 419–424.
- [206] O. Burdakov, P. Doherty, K. Holmberg, and P.-M. Olsson, "Optimal placement of UV-based communications relay nodes," J. Glob. Optim., vol. 48, pp. 511–531, Dec. 2010.
- [207] D. Tsonev, S. Videv, and H. Haas, "Light fidelity (Li-Fi): Towards all-optical networking," in *Proc. 8th Broadband Access Commun. Technol.*, 2014, Art. no. 900702.
- [208] S. Yu, X. Wu, Y. Wang, X. Guo, and L. Tong, "2D materials for optical modulation: Challenges and opportunities," *Adv. Mater.*, vol. 29, no. 14, 2017, Art. no. 1606128.
- [209] K. C. Balram, M. I. Davanço, B. R. Ilic, J.-H. Kyhm, J. D. Song, and K. Srinivasan, "Acousto-optic modulation and optoacoustic gating in piezo-optomechanical circuits," *Phys. Rev. Appl*, vol. 7, no. 2, 2017. Art. no. 24008.

- [210] D. P. Young, J. Brewer, J. Chang, T. Chou, J. Kvam, and M. Pugh, "Diffuse mid-UV communication in the presence of obscurants," in *Proc. 46th Conf. Rec. Asilomar Conf. Signals, Syst. Comput. (ASILOMAR)*, 2012, pp. 1061–1064.
- [211] X. Sun et al., "71-Mbit/s ultraviolet-B LED communication link based on 8-QAM-OFDM modulation," *Opt. Exp.*, vol. 25, no. 19, pp. 23267–23274, 2017.
- [212] G. A. Mahdiraji and E. Zahedi, "Comparison of selected digital modulation schemes (OOK, PPM and DPIM) for wireless optical communications," in *Proc. 4th Student Conf. Res. Develop.*, 2006, pp. 5–10.
- [213] H. Wu and Q. Fan, "Study on LED visible light communication channel model based on poisson stochastic network theory," in *Proc. Int. Conf. Wireless Commun. Smart Grid (ICWCSG)*, 2020, pp. 5–9.
- [214] H. Park and J. R. Barry, "Modulation analysis for," in *Proc. IEEE Int. Conf. Commun. ICC*, 1995, pp. 1182–1186.
- [215] L. Ahlin, J. Zander, and S. Ben Slimane, *Principles of Wireless Communications*. Lund, Sweden, Studentlitteratur, 2006.
- [216] J. M. Kahn and J. R. Barry, "Wireless infrared communications," Proc. IEEE, vol. 85, no. 2, pp. 265–298, Feb. 1997.
- [217] J. Zhang, "Modulation analysis for outdoors applications of optical wireless communications," in *Proc. Int. Conf. Commun. Technol. Proc. (ICCT)*, 2000, pp. 1483–1487.
- [218] Z. F. Ghassemlooy, R. U. Reyher, E. Kaluarachchi, and A. Simmonds, "Digital pulse interval and width modulation for optical fiber communications," in *Proc. All-Opt. Commun. Syst. Archit., Control, Netw. Issues*, 1995, pp. 60–68.
- [219] D.-S. Shiu and J. M. Kahn, "Differential pulse-position modulation for power-efficient optical communication," *IEEE Trans. Commun.*, vol. 47, no. 8, pp. 1201–1210, Aug. 1999.
- [220] W. O. Popoola, E. Poves, and H. Haas, "Spatial pulse position modulation for optical communications," *J. Lightw. Technol.*, vol. 30, no. 18, pp. 2948–2954, Sep. 15, 2012.
- [221] Y. Jiao, Y. Zhao, L. Kuang, R. Lin, J. Ning, and J. Liu, "Design and optimization of an ultraviolet scattering communication system based on duty cycle regulation," *Photonics*, vol. 11, no. 7, p. 662, 2024.
- [222] K. Kiasaleh, "Performance of APD-based, PPM free-space optical communication systems in atmospheric turbulence," *IEEE Trans. Commun.*, vol. 53, no. 9, pp. 1455–1461, Sep. 2005.
- [223] Z. Zhou, B. Vucetic, M. Dohler, and Y. Li, "MIMO systems with adaptive modulation," *IEEE Trans. Veh. Technol.*, vol. 54, no. 5, pp. 1828–1842, Sep. 2005.
- [224] N. D. Chatzidiamantis, A. S. Lioumpas, G. K. Karagiannidis, and S. Arnon, "Adaptive subcarrier PSK intensity modulation in free space optical systems," *IEEE Trans. Commun.*, vol. 59, no. 5, pp. 1368–1377, May 2011.
- [225] N. Alshaer, T. Ismail, and M. E. Nasr, "Enhancing earth-to-satellite FSO system spectrum efficiency with adaptive M-ary PSK and SIMO in presence of scintillation and beam wander," *AEU Int. J. Electron. Commun.*, vol. 125, Oct. 2020, Art. no. 153366.
- [226] X. He et al., "1 Gbps free-space deep-ultraviolet communications based on III-nitride micro-LEDs emitting at 262 nm," *Photon. Res.*, vol. 7, no. 7, pp. B41–B47, 2019.
- [227] O. Alkhazragi et al., "2.4-Gbps ultraviolet-C solar-blind communication based on probabilistically shaped DMT modulation," in *Proc. Opt. Fiber Commun. Conf. Exhib.*, 2020, pp. 1–3.
- [228] S. Zhu et al., "2 Gbps free-space ultraviolet-C communication based on a high-bandwidth micro-LED achieved with preequalization," Opt. Lett., vol. 46, no. 9, pp. 2147–2150, 2021.
- [229] J. J. D. McKendry et al., "4 Gbps wireless optical communications up to 5 m using a UV-C micro-light-emitting diode array," in *Proc. IEEE Photon. Conf. (IPC)*, 2021, pp. 1–2.
- [230] D. M. Maclure et al., "Gb/s optical wireless communications up to 17 meters using a UV-C micro-light-emitting diode," in *Proc. IEEE Photon. Conf. (IPC)*, 2022, pp. 1–2.
- [231] Z. Qian et al., "Size-dependent UV-C communication performance of AlGaN micro-LEDs and LEDs," J. Lightw. Technol., vol. 40, no. 22, pp. 7289–7296, Nov. 15, 2022.
- [232] Y. Yoshida et al., "Gbps-class solar-blind WDM optical wireless communication by (264, 274, 282)-nm deep-UV LEDs and CsTe photomultiplier tube," in *Proc. Opt. Fiber Commun. Conf. Exhib.* (*OFC*), 2023, pp. 1–3.

- [233] D. M. Maclure et al., "Hundred-meter Gb/s deep ultraviolet wireless communications using AlGaN micro-LEDs," *Opt. Exp.*, vol. 30, no. 26, pp. 46811–46821, 2022.
- [234] L. Gupta, R. Jain, and G. Vaszkun, "Survey of important issues in UAV communication networks," *IEEE Commun. Surveys Tuts.*, vol. 18, no. 2, pp. 1123–1152, 2nd Quart., 2016.
- [235] C. Yan, L. Fu, J. Zhang, and J. Wang, "A comprehensive survey on UAV communication channel modeling," *IEEE Access*, vol. 7, pp. 107769–107792, 2019.
- [236] M. A. El-Shimy and S. Hranilovic, "Binary-input non-line-of-sight solar-blind UV channels: Modeling, capacity and coding," J. Opt. Commun. Netw., vol. 4, no. 12, pp. 1008–1017, Dec. 2012.
- [237] S. Arya and Y. H. Chung, "Novel indoor ultraviolet wireless communication: Design implementation, channel modeling, and challenges," *IEEE Syst J.*, vol. 15, no. 2, pp. 2349–2360, Jun. 2021.
- [238] O. Alkhazragi et al., "Gbit/s ultraviolet-C diffuse-line-of-sight communication based on probabilistically shaped DMT and diversity reception," *Opt. Exp.*, vol. 28, no. 7, pp. 9111–9122, 2020.
- [239] Q. Tian et al., "Turbo-coded 16-ary OAM shift keying FSO communication system combining the CNN-based adaptive demodulator," *Opt. Exp.*, vol. 26, no. 21, pp. 27849–27864, 2018.
- [240] C. Jia, P. Wang, Y. Li, C. Huang, H. Fu, and W. Pang, "ABERs of LDPC-coded multi-hop FSO over double GG fading channels with pointing error and path loss," *IEEE Photon. Technol. Lett.*, vol. 30, no. 15, pp. 1357–1360, Aug. 1, 2018.
- [241] X. Liang, M. Zhang, and D. Han, "Security performance of polar codes in uv wireless communications," in *Proc. Asia Commun. Photon. Conf. (ACP)*, 2018, pp. 1–3.
- [242] M. Wu, D. Han, X. Zhang, F. Zhang, M. Zhang, and G. Yue, "Experimental research and comparison of LDPC and RS channel coding in ultraviolet communication systems," *Opt. Exp.*, vol. 22, no. 5, pp. 5422–5430, 2014.
- [243] Q. An et al., "Comparison of LDPC and turbo codes in ultraviolet wireless communication systems," in *Proc. Asia Commun. Photon. Conf. (ACP)*, 2015, Art. no. ASu2A-82.
- [244] W. Hu, Z. Luo, D. Han, Q. Chen, L. Ai, and M. Zhang, "A scheme of ultraviolet communication system with polar channel coding," in *Proc. 16th Int. Conf. Opt. Commun. Netw. (ICOCN)*, 2017, pp. 1–3.
- [245] Y. Zhang, X. Zhao, X. Zhao, H. Zhang, and D. Han, "Performance analysis of LDPC codes with ID filter in NLOS UV communication system," in *Proc. Asia Commun. Photon. Conf. (ACP)*, 2013, Art. no. AF2F-50.
- [246] Y. Peng and D. Zhou, "Performance of UV spread spectrum communication system based on LDPC codes," in *Proc. 13th Int. Symp. Comput. Intell. Design (ISCID)*, 2020, pp. 131–135.
  [247] W. Wang, T. Wu, G. Yao, J. Ma, and P. Su, "The bit error rate
- [247] W. Wang, T. Wu, G. Yao, J. Ma, and P. Su, "The bit error rate performance analysis of LDPC code in ultraviolet communication system," in *Proc. 10th Opt. Design Testing*, 2020, pp. 339–346.
- [248] H. Qin, Y. Zuo, D. Zhang, Y. Li, and J. Wu, "Received response based heuristic LDPC code for short-range non-line-of-sight ultraviolet communication," *Opt. Exp.*, vol. 25, no. 5, pp. 5018–5030, 2017.

- [249] W. Wang, P. Su, and J. Ma, "LDPC channel coding in ultraviolet communication system," in *Proc. 18th Int. Conf. Opt. Commun. Netw.* (ICOCN), 2019, pp. 1–3.
- [250] A. Du et al., "Performance analysis of LDPC coding under multiple scattering in ultraviolet communication," in *Proc. 20th Int. Conf. Opt. Commun. Netw. (ICOCN)*, 2022, pp. 1–3.
- [251] L. Ai, M. Zhang, D. Han, W. Hu, X. Liang, and Z. Ghassemlooy, "Investigating the optimal SCL decoding algorithm for polar codes in ultraviolet communication," in *Proc. 11th Int. Symp. Commun. Syst., Netw. Digit. Signal Process. (CSNDSP)*, 2018, pp. 1–5.
- [252] W. Hu et al., "Research on channel-related polar code with an optimum code length for wireless ultraviolet communications," *Opt. Exp.*, vol. 25, no. 23, pp. 28630–28642, 2017.
- [253] T.-f. Zhao, Y. Xie, X. Liu, and P. Song, "Research on novel fountain code for UAV formation flight control in UV communication," in Proc. 7th IEEE Int. Symp. Microw, Antenna, Propagat., EMC Technol. (MAPE), 2017, pp. 120–123.
- [254] R. Li, T. Chen, L. Fan, and A. Dang, "Performance analysis of a multiuser dual-hop amplify-and-forward relay system with FSO/RF links," J. Opt. Commun. Netw., vol. 11, no. 7, pp. 362–370, Jul. 2019.
- [255] S. Song, Y. Liu, T. Xu, and L. Guo, "Hybrid FSO/RF system using intelligent power control and link switching," *IEEE Photon. Technol. Lett.*, vol. 33, no. 18, pp. 1018–1021, Sep. 15, 2021.
- [256] H. Safi, A. A. Sharifi, M. T. Dabiri, I. S. Ansari, and J. Cheng, "Adaptive channel coding and power control for practical FSO communication systems under channel estimation error," *IEEE Trans. Veh. Technol.*, vol. 68, no. 8, pp. 7566–7577, Aug. 2019.
- [257] M. Karimi and M. Uysal, "Novel adaptive transmission algorithms for free-space optical links," *IEEE Trans. Commun.*, vol. 60, no. 12, pp. 3808–3815, Dec. 2012.
- [258] S. H. Hariq, N. Odabasioglu, and M. Uysal, "An adaptive modulation scheme for coded free-space optical systems," in *Proc. 19th Eur. Conf. Netw. Opt. Commun. (NOC)*, 2014, pp. 132–135.
- [259] L. Liu, M. Safari, and S. Hranilovic, "Rate-adaptive FSO communication via rate-compatible punctured LDPC codes," in *Proc. IEEE Int. Conf. Commun. (ICC)*, 2013, pp. 3948–3952.
- [260] J. A. Anguita, M. A. Neifeld, B. Hildner, and B. Vasic, "Rateless coding on experimental temporally correlated FSO channels," *J. Lightw. Technol.*, vol. 28, no. 7, pp. 990–1002, Apr. 2010.
- [261] M. S. Salah, W. G. Cowley, and K. D. Nguyen, "Adaptive transmission schemes for free-space optical channel," in *Proc. 8th Int. Conf. Signal Process. Commun. Syst. (ICSPCS)*, 2014, pp. 1–7.
- [262] I. B. Djordjevic, "Adaptive modulation and coding for free-space optical channels," J. Opt. Commun. Netw., vol. 2, no. 5, pp. 221–229, May 2010.
- [263] F. Long, J. Tang, L. Li, J. Chen, C. Gong, and Z. Xu, "Adaptive modulation and coding with LDPC codes and retransmission for ultraviolet communication," *IEEE Photon. J.*, vol. 16, no. 3, pp. 1–10, Jun. 2024.
- [264] H. Wu, "Architecture, implementation, and experimentation of ultraviolet communication systems," Ph.D. dissertation, Electr. Eng., Univ. British Columbia, Vancouver, BC, Canada, 2024.